

June 26, 2002

http://caster.ssw.upenn.edu/~restes/CSEC_Files/CSEC_Bib_August_2001.pdf

42 pg

PREFACE

A. Introduction

Child sexual abuse (CA) and *child sexual exploitation* (CSE) surfaced in the last decade of the 20th century as two of the most neglected forms of child abuse. A third form of child abuse--the *commercial* sexual exploitation of children (CSEC)--appears to be an even more insidious and, daily, affects the lives of tens of thousands of children in the United States, and tens of millions of children worldwide.¹ Some portion of these children are in the "employment" of well organized networks of traffickers in child and adult sex, at least some of which also engage in the sale of illegal drugs, money laundering activities and other criminal activities.

Patterns of child sexual exploitation appear to be fueled by: 1) the use of "survival sex" by runaway and thrownaway children to provide for their subsistence needs (Flowers, 1994; Greene et al., 1999; Haq, 1996; Johnson et al., 1996; Kral, 1997; Yates, 1991; Whitbeck & Simons, 1990); 2) the presence of pre-existing adult prostitution markets in communities where large numbers of street youth are concentrated (Farley & Kelly, 2000; Flowers, 1994; Hofstede, 1999); 3) prior history of child sexual abuse and child sexual assault (Briere, 1998; McClanahan, 1999; Mullen, 1996; Powers & Jaklitsch, 1989; Rotheram-Borus, et al., 1996; Seng, 1989; Simons & Whitbeck, 1991; Widom & Kuhns, 1996); 4) poverty (Azaola, 2001; Hood-Brown, 1998; Lederer, 1996; Longford, 1995; Unicef, 2000); 5) the presence of large numbers of unattached and transient males in local communities--including military personnel, truckers, and conventioneers among others (Moon, 1997; Sturdevant et al., 1992); 6) for some girls, membership in gangs (Hazelhurst & Hazelhurst, 1998; Moore & Hagedorn, 2001); 7) the promotion of child prostitution by parents, older siblings and boy friends (Dembo et al., 1992; Faugier & Sargent, 1997; Mueck, 1992); 8) the recruitment of children as "sex workers" by organized crime units (Budapest Group, 1999; Harris, 1998a; Lanning, 1992; Williams, 1995); and, increasingly, 9) illegal trafficking of children for sexual purposes both within and to the U.S. from developing countries located in Asia, Africa, Central and South America, and Central and Eastern Europe (Barr et al., 1996; Hughes, 2000b; Knight, 1998; Lederer, 2001; Miko & Park, 2000).

B. <u>Recruitment of Children Into SEC</u>

The processes whereby children are "recruited" into SEC are varied and are complex. They nearly always involve adult accomplices—including parents and older siblings. According to the Exploited Child Unit (hereafter "ECU") of the National Center for Missing and Exploited Children (1998),

Child prostitutes are victims of sexual abuse. Many of them have run away from homes where they have been sexually or physically abused. Some come from families who no longer want them or who feel they can no longer handle them. These children often see themselves as their only supporter. Under such circumstances, some fall into prostitution as a way to survive or as a way to get the things they want or need...Unfortunately, these victims often become involved in the criminal-justice system as offenders. In order to support themselves, or to escape the life that they have come to lead, they get involved in the use and/or sale of drugs, theft, or robberies.

¹ The precise number of sexually exploited children is unknown. However, the United Nations Children's Fund estimates that more than a hundred million children worldwide are employed as "sex workers," i.e., as prostitutes, as subjects of pornography, or both (Unicef, 1997). In addition, hundreds of thousands of children shuttle the planet each year as part of a well-concealed network operated by international traffickers in children for sex. The number of sexually exploited children in the United States also is unknown but conservative estimates place their numbers at between 300,000 and 500,000.

Other young people are recruited into "sex work" through forced abduction (Barr et al., 1996; DoL, 1995), by pressure from their parents (D'Asaro & Foley, 1997), through deceptive agreements between parents and traffickers in the CSEC, including unrecognized representatives of crime rings (Barr et al. 1996; D'Asaro & Foley, 1997; DoL, 1995; Miko & Park, 2000; Seabrook, 1997; Yoon, 1997). Once recruited, these children typically are taken or travel to "work sites" located great distances from their place of origin. Isolation from their families and friends is the norm. Few are able to establish new relationships with persons other than those who are "managing" their victimization. Violence, forced drug use and threats to the point of death are only part of the daily abuse to which the majority of these children and youth are subjected.

Poverty is the most frequent explanation cited for the involvement of large numbers of children in sex crimes (Azaola, 2001a; Boye, 1996; Longford, 1995; Mayombo, 1998; Save the Children, 1996; Shamim, 1993).² But as suggested by the ECU, poverty alone does not account for the large number of children under the age of 16 years being recruited into the sex industry, especially in rich countries such as the U.S. and Canada (Dionne, 2001; Estes & Weiner, 2001; Tremblay, 2001). Indeed, many children recruited into prostitution in the U.S. and Canada appear to come from middle class families and engage in "survival sex" or prostitution as way of supporting themselves on the nation's streets (Green, 1999; Hunnicutt, 2001; NCMEC, 1998; Snell, 1995; Stiffman, 1989).

Other powerful explanations of the CSEC include *pedophilia* (Cole, 1993; De Mause, 1991; Knight, Carter, & Prentky, 1989; Prentky & Knight, 1993; Prentky, Knight, & Lee, 1997), *ease of access* (Harris, 1998a, 1998c; O'Grady, 1992), *relaxed legal enforcement environments* (Editor, 1996; Gutierrez, 1998; Harris, 1997a; Hodgson, 1995; Samath, 1998), *debt bondage* (DoL, 1996; Knight, 1998; Youth Plus, 1995), *sadomasochism* (Finkelhor & Brown, 1985), *inter-generational prostitution* (Seneviratne, 1994), and the *high remittances* received by child sending/sex tourist-receiving countries (Barr et al. 1996; Boye, 1996; ECPAT, 1996a; IBCR, 1998a, 1998b). Other explanations for the CSEC also are plausible but, to date, they have been less fully studied: *high economic demand*, which stimulates the supply of children into the sex trade (Barr et al. 1996; Ennew, 1996; Farley, 1998; Hughes, 2000a; Yoon, 1997); community disintegration (D'Asaro & Foley, 1997; Dembo, 1992); *social and cultural devaluation* of children (Flowers, 1994; Gutierrez, 1998); and, pre-existent *international crime organizations* with transnational transportation and financial capabilities (Barr et al. 1996; DoL, 1996; Muntarbhorn, 1996; Williams, 1995; Yoon, 1997). Within this array of potentially influential factors, of course, there also needs to be a convergence in available children, traffickers, and customers.

C. The Impact of Sexual Exploitation on Children

However they fall victims of sexual exploitation, few children are able to escape their molestation unharmed; virtually all suffer long term physical and emotional injures (Barnett, Manly & Cicchetti, 1993; Cevallos, 1998; Finkelhor & Brown, 1985; Gelles, 1994, 1998; Goldstein, 1987; Kilpatrick & Saunders, 1997; Straus & Gelles, 1988; Vittachi, 1989). These injuries remain with the children throughout adolescence and adulthood and, in turn, pose complex service challenges for the justice (Bala & Schwartz, 1993; Collins, Schwartz, & Epstein, 1998; Schwartz, 1997; Schwartz, Fishman, and & Vleet, 1996) and human service systems (Fishman, Schwartz & Hsieh, 1997; Schwartz & Au Claire, 1995). The relationship is known to be especially strong between child sexual victimization and *teen pregnancy* (Ireland & Widom, 1994; Smith & Thornberry, 1995; Widom, 1996; Widom, & Kuhns, 1996), *adult prostitution* (Widom & Kuhns, 1996), substance abuse (Ireland & Widom, 1994; Kelley, Thornberry & Smith, 1997),

² For discussions of the highly unfavorable social, political, economic, and legal conditions that contribute to the CSEC in other world regions see Estes, 1995, 1996, 1997, 1998a, 1998b.

violence (Gelles & Wolfner, 1994; Schwartz, Rendon, & Hsieh, 1994; Weiner, 1994; Weiner & Ruback, 1998; Weiner & Wolfgang, 1989), among other forms of adult criminal behavior (Kelley, Thornberry & Smith, 1997; Loeber & Farrington, 1998; Smith & Thornberry, 1995).

D. The Emerging Statistical Picture: A Cause For Concern

Though estimates vary concerning the number of sexually exploited children, the United Nation's Children's Fund (UNICEF) believes their numbers to exceed 100 million worldwide, not all of whom are located in "poor" or "developing" countries (UNICEF, 1997). Indeed, the first *World Congress Against the Commercial Sexual Exploitation of Children* (Muntarbhorn, 1996) confirmed that large numbers of child "sex workers" are to be found in rich countries, including in the U.S. for which the "End Child Prostitution, Child Pornography, and the Trafficking of Children for Sexual Exploitation" (ECPAT) estimated their numbers to be between 100,000 and 300,000 (ECPAT, 1996b:70). Other estimates of child sexual exploitation suggest the numbers of such children to be even higher (Goldman & Wheeler, 1986; Greenfeld, 1997; Spangenberg, 2001).

Certainly, experts in child sexual exploitation believe the numbers of such children in the United States to be substantial (Greene et al., 1999; Hughes & Roche, 1999b; Kral et al., 1997). America's problems with child sexual exploitation is widely believed to be concentrated in cities and towns that border Mexico (Azaola, 2001a; Harris, 1998a; Knight, 1998) and Canada (Dionne, 2001; Hecht, 1997). However, the problem also is believed to be extensive in America's coastal states where large numbers of children from Asia, Africa, Latin America and Eastern Europe enter the country illegally (DoL, 1996). Organized crime groups also appear to be involved in child pornography and child prostitution in the U.S., but especially criminal organizations with ties to Russia, the Ukraine, and other countries of the Former Soviet Union (Budapest Group, 1999; Hughes, 2000b; INTERPOL, 1997; Richard, 1999). Crime groups with ties to China, the Philippines, and Thailand also are known to be involved in trafficking children for sexual purposes into the United States (Biotti, 2000; ECPAT, 1996a; Richard, 1999).

Bibliography on

THE SEXUAL EXPLOITATION OF CHILDREN

The bibliography that follow was designed to introduce readers to the very rich theoretical and empirical literatures relating to *child abuse, child sexual abuse, child sexual exploitation*, and the *commercial sexual exploitation of children*. The bibliography also contains introductory references to the international child rights movement, including worldwide efforts to protect children from sexual exploitation. Throughout, the bibliography contains references to research of an international and comparative nature. Charts 1 & 2 define the major concepts used in preparing the bibliography.

Sexual	Definitions of Terms Associated with the Sexual Exploitation (SEC) and Commercial Sexual Exploitation of Children (CSEC)		
Concept	Subtypes	Definition	
Child		Persons under the age of 18 years unless, under the law applicable to the child, majority is attained earlier (United Nations Convention on the Rights of the Child, 1989: Article 1)	
Child Abuse		The recurrent infliction of physical or emotional injury on a depend- ent minor, through intentional beatings, uncontrolled corporal pun- ishment, persistent ridicule and degradation, or sexual abuse, usually committed by parents or guardians (National Association of Social Workers, 1994).	
Child Sexual Abuse (CSA)		Sexual activity involving persons younger than 18 years of age. Most often perpetrated by an adult, such activities include rape and molestation, pornography, and exposure of children to the sexual acts of others (ala National Clearinghouse on Child Abuse and Ne- glect Information, 1996).	
	Type I Rape and Molestation (CSA-1)	The carnal knowledge of a person, forcibly and/or against that per- son's will; or not forcibly or against the person's will where the vic- tim is incapable of giving consent because of his/her temporary or permanent mental or physical incapacity. (National Incident-Based Reporting System [NIBRS] as cited in Snyder 2000:13).	
	Type 2 Pornography (CSA-2)	The employment, use, persuasion, inducement, enticement, or coer- cion of any child to engage in, or assist any other person to engage in, any sexually explicit conduct or simulation of such conduct for the purpose of producing a visual depiction of such conduct (Na- tional Clearinghouse on Child Abuse and Neglect Information, 1996)	
	Type 3 Exposure of Children to the Sexual Acts of Others (CSA-3)	The intentional exposure of children to sexual acts performed by others (including those engaged in by parents, caregivers and others entrusted with the care and nurturance of children)	

Chart 1 Definitions of Terms Associated With the tion (SEC) and Commercial Sexual Exploitation of Childu

Concept	Subtypes	Definition
Sexual		Any sexual act directed against a person forcibly and/or against that
Assault		person's will; or not forcibly or against the person's will where the
(SAs)		victim is incapable of giving consent because of his/her temporary
		or permanent mental or physical incapacity (ala National Incident-
		Based Reporting System [NIBRS] as cited in Snyder 2000:13).
	Type 1	The carnal knowledge of a person, forcibly and/or against that per-
	Forcible Rape	son's will; or not forcibly or against the person's will where the vic-
	(SA-1)	tim is incapable of giving consent because of his/her temporary or
	(5A-1)	permanent mental or physical incapacity. (National Incident-Based
		Reporting System [NIBRS] as cited in Snyder 2000:13).
		Reporting System [MBRS] as cried in Siryder 2000.15).
		If force was used on threatened, the prime is clearlifted as "forcible
		If force was used or threatened, the crime is classified as "forcible
		rape" regardless of the age of the victim. If no force was used or
		threatened and the victim is under the statutory age of consent, the
		crime is classified as "statutory rape" (National Incident-Based Re-
		porting System [NIBRS] as cited in Snyder, 2000:13).
	Type 2	Oral or anal sexual intercourse with another person, forcibly and/or
	Forcible Sodomy	against that person's will; or not forcibly or against the person's will
	(SA-2)	where the victim is incapable of giving consent because of his/her
	· ·	youth or because of his/her temporary or permanent mental or
		physical incapacity (National Incident-Based Reporting System
		[NIBRS] as cited in Snyder 2000:13).
	Type 3	To use an object or instrument to unlawfully penetrate, however
	Sexual Assault With	slightly, the genital or anal opening of the body of another person,
	An Object	forcibly and/or against that person's will; or not forcibly or against
	(SA-3)	the person's will where the victim is incapable of giving consent
	(5/1 5)	because of his/her youth or because of his/her temporary or perma-
		nent mental or physical incapacity (National Incident-Based Report-
		ing System [NIBRS] as cited in Snyder 2000:13).
		An "chicot" or "instrument" is enuthing used by the offender other
		An "object" or "instrument" is anything used by the offender other
		than the offender's genitalia (National Incident-Based Reporting
		System [NIBRS] as cited in Snyder 2000:13).
	Type 4	The touching of the private body parts of another person for the
	Forcible Fondling	purpose of sexual gratification. Forcibly and/or against that person's
	(SA-4)	will; or not forcibly or against the person's will where the victim is
		incapable of giving consent because of his/her youth or because of
		his/her temporary or permanent mental or physical incapacity.
		Forcible fondling includes "indecent liberties" and "child molest-
		ing" (National Incident-Based Reporting System [NIBRS] as cited
		in Snyder 2000:13).
Child Sexual		Any sexual act directed against a person younger than 18 years of
Assault		age, forcibly and/or against that person's will; or not forcibly or
(CSAs)		against the person's will where the victim is incapable of giving
(00.10)		consent because of his/her temporary or permanent mental or physi-
		cal incapacity (ala National Incident-Based Reporting System
		[NIBRS] as cited in Snyder 2000:13).
		See definitions listed under "Sexual Assault"
		SA-1: Forcible Rape
		SA-2: Forcible Sodomy
		SA-3: I Sexual Assault With An Object
		SA-4: Forcible Fondling
Customer	1	A person who buys goods or services. The purchase may be made

Page 5 of 42

Concept	Subtypes	Definition
		using cash, in-kind exchanges, or on the basis of a promise to pay for the desired goods or services at some future time.
Pornography		Films, videos, magazines, writings, photographs, computer images, or other materials that are sexually explicit and intended to cause sexual arousal in the viewer.
Child Pornography (CP)		Films, videos, magazines, writings, photographs, computer image, or other materials produced by either adults or children, or both, that contain sexually explicit images of children and youth under the age of 18 years.
		Child pornography often has considerable commercial value; typi- cally, though, child pornography is "traded" or exchanged between pedophiles rather than sold.
	<i>Type 1</i> Images Depicting Children Only (CP-1)	Films, videos, magazines, writings, photographs, computer image, or other materials produced by adults that contain only sexually explicit images of children and youth under the age of 18 years.
	Type 2 Images Depicting Children With Adult(s) (CP-2)	Films, videos, magazines, writings, photographs, computer image, or other materials produced by adults that contain sexually explicit images of children, youth and adults.
	<i>Type 3</i> Images Depicting Juveniles With Juveniles (CP-3)	Films, videos, magazines, writings, photographs, computer image, or other <i>materials produced and distributed by youth under the age</i> of 18 years that contain sexually explicit images of children and youth together.
Prostitution		The act of engaging in sexual intercourse or performing other sex acts in exchange for money or other considerations (e.g., food, clothing shelter, affection, etc.).
Child/Juvenile Prostitution (CPR)		The act of engaging in sexual intercourse or performing other sex acts with a child in exchange for money, clothing, food, shelter, drugs, or other considerations (World Health Organization, 1996).
Exploitation		Unfair, if not illegal, treatment or use of somebody or something, usually for personal gain.
Sexual Exploitation (SE)		A practice by which a person achieves sexual gratification, financial gain or advancement through the abuse or exploitation of a person's sexuality by abrogating that person's human right to dignity, equality, autonomy, and physical and mental well-being; i.e. trafficking, prostitution, prostitution tourism, mail-order-bride trade, pornography, stripping, battering, incest, rape and sexual harassment (Hughes, 1999).
Child Sexual Exploitation (CSE)	۲	A practice by which a person, usually an adult, achieves sexual gratification, financial gain or advancement through the abuse or exploitation of a child's sexuality by abrogating that child's human right to dignity, equality, autonomy, and physical and mental wellbeing, i.e. trafficking, prostitution, prostitution tourism, mail-order- bride trade, pornography, stripping, battering, incest, rape and sexual harassment (ala Hughes, 1999).
		CSE reflects a continuum of abuse ranging from child sexual abuse to child sexual exploitation to the commercial sexual exploitation of children.

Page 6 of 42

Concept	Subtypes	Definition
	Type 1 Child Sexual Abuse	See definition listed under "Child Sexual Abuse" CSA-1: Rape and Molestation
	(CSA)	CSA-2: Pornography CSA-3: Exposure of Children to the Sexual Acts of Others
·	<i>Type 2</i> Child Sexual Assault	See definition listed under "Child Sexual Assault" CSAs-1: Forcible Rape
	(CSAs)	CSAs-2: Forcible Sodomy
		CSAs-3: Sexual Assault With An Object CSAs-4: Forcible Fondling
	<i>Type 3</i> The Commercial Sex-	See definitions listed under the "Commercial Sexual Exploitation of Children"
	ual Exploitation of	CSEC-1: Child Pornography
	Children	CSEC-2: Child/Juvenile Prostitution
	(CSEC)	CSEC-3: Trafficking in Children For Sexual Purposes
Commercial		The sexual exploitation of children of children (SEC) entirely, or at
Sexual Exploi-		least primarily, for financial or other economic reasons. The eco-
tation of		nomic exchanges involved may be either monetary or non-monetary
Children (CSEC)	1	(i.e., for food, shelter, drugs) but, in every case, involves maximum
(CSEC)		benefits to the exploiter and an abrogation of the basic rights, dig- nity, autonomy, physical and mental well-being of the children in-
		volved.
	Type I	See definitions listed under "Child Pornography"
`	Child	CP-1: Images Depicting Children Only
	Pornography	CP-2: Images Depicting Children With Adults
	(CSEC-1)	CP-3: Images Depicting Juveniles With Juveniles
	Type 2	See definitions listed under "Child Prostitution" and "Survival Sex"
	Child/Juvenile	See definitions listed under Cana Prostitution and Survival Sex
	Prostitution	
	(CSEC-2)	
	Туре 3	See definitions listed under "Trafficking"
	Trafficking in Chil-	T-1: Domestic Trafficking
	dren For Sexual Pur-	T-2: International Trafficking
	poses (CSEC-3)	
Tourist	(0020 0)	A person or persons who visit places away from home for pleasure.
Sex Tourist		Persons who travel from their homes, usually across international
		borders, with the intent of engaging in sexual activities with others,
C T		including children.
Sex Tourism		Commercially organized travel and related services (e.g., hotel, transportation), usually across international borders, for persons
		seeking to engage in sex with citizens of other countries, including
		children who are citizens of those countries.
"Survival Sex"		Many youth involved in the exchange of sex for money or other
		considerations (e.g., food, shelter, drugs, etc.) do not perceive them-
		selves as engaging in prostitution but rather as doing "whatever is
		necessary" to ensure their survival. For purposes of this study,
		however, "survival sex" and "child prostitution" are understood to
	ļ	be the same phenomenon and the terms are used interchangeably.
Trafficking		The transport, harboring, or sale of persons within national or across
<i>(T</i>)	1	international borders through coercion, force, kidnapping, deception
		or fraud, for purposes of placing persons in situations of forced la-
		bor or services, such as forced prostitution, domestic servitude, debt bondage or other slavery-like practices. Agreement exists that the
	l	bondage of other stavery-like practices. Agreement exists that the

Concept	Subtypes	Definition
'		concept applies whether a child was taken forcibly or voluntarily (18 USC 1589 et seq.).
	Type 1	The recruitment, transportation or receipt of children through decep-
	Domestic	tion or coercion for the purpose of prostitution, other sexual exploi-
	Trafficking	tation or forced labor only within their own country. Children may
	(T-1)	be trafficked domestically either voluntarily or involuntarily.
`		a. Voluntary domestic trafficking involves the movement of chil-
		dren voluntarily across state lines for the purpose of bringing
		financial gain to either the children or the traffickers, or both.
		The majority of children trafficked in this way are required ei-
		ther to pay fees to or to perform services, including sexual services, for their traffickers.
		b. Involuntary domestic trafficking involves the movement of chil-
		dren involuntarily across intra-national (e.g., state) lines for the purpose of bringing financial gain to the traffickers. The
		majority of children trafficked are held in servitude and are
		forced to pay trafficking fees through a combination of inden-
		tured services, including commercial sexual services.
	Type 2	The recruitment, transportation or receipt of children through decep-
	International	tion or coercion for the purpose of prostitution, other sexual exploi-
	Trafficking	tation or forced labor across international boundaries. Children
	(T-2)	may be trafficked internationally either voluntarily or involuntarily.
		a. Voluntary international trafficking involves the movement
		of children voluntarily across international borders for the
,		purpose of bringing financial gain to either the children or
		the traffickers, or both. The majority of children trafficked
		in this way are required either to pay fees or to perform ser-
		vices, including sexual services, to their traffickers.
		b. Involuntary international trafficking involves the movement
		of children involuntarily across international borders for the
		purpose of bringing financial gain to the traffickers. The
		majority of children trafficked internationally are held in
		servitude and are forced to pay trafficking fees through a
		combination of indentured services, including commercial sexual services.
Smuggling		The procurement of illegal entry of a person into a State of which
		the latter person is not a national with the objective of making a profit (United Nations, 1999:3).
		Smugaling is distinguished from trafficking in that align amugaling
		Smuggling is distinguished from trafficking in that alien smuggling
		involves the provision of a service, albeit illegal, to people who knowingly buy the service in order to get into a foreign country.
Organized	•	A non-ideological enterprise involving a number of persons in close
Organizea Crime		social interaction, organized on a [structured] basis with [different]
		levels/ranks, for the purpose of securing profit and power by engag-
		ing in illegal and legal activities (Abadinsky, 1994:8 as cited in
		Schloenhardt, 1999:9 and Graycar, 1999:7-8).
Transnational	• •	The crossing of a border by people, things or criminal will, together
Crime		with the international recognition of the crime at both national and
		international levels (Secretary-General of INTERPOL as cited by

Concept	Subtypes	Definition
	-	Graycar, 1999:2)
	. ·	To be considered "international," a crime must be a criminal offense in at least two nation states, thereby bringing into effect interna- tional conventions, extradition treaties or concordant national laws (Bossard, 1990:5 as cited in McFarlane, 1999:2 and Graycar, 1999:2-3).

Sele	ected Terms Relating to the Sexual Exploitation of Children (SEC)
Concept	Definition
Call Boy/ Call Girl	A prostitute (male/female) who responds to telephone calls for sex. In all cases, call boys and call girls travel to the prospective client's residence, hotel or other desig- nated meeting spot.
Exhibitionist	A person who repeatedly exposes their genitals to unsuspecting strangers in order to achieve sexual excitement.
Hebephile	An adult with sexual desires and arousal fantasies that often culminate in sexual acts with pubescent children of the same or opposite sex.
Hustler	A prostitute, especially a streetwalker or one who solicits in bars (slang).
"John"	A man who is a prostitute's customer (slang).
Pander	To procure sexual favors for somebody. (Synonym: solicit, procure)
Pederast	Men with sexual desires and arousal fantasies that often culminate in sexual acts with pre-pubescent boys.
Pedophile	An adult with sexual desires and arousal fantasies that often culminate in sexual acts with pre-pubescent children of the same or opposite sex.
Pimp	One who promotes and/or profits from the sale and/or abuse of another person's body or sexuality for sexual purposes, or the production and/or sale images made of that person, e.g. trafficker, pornographer, brothel madam, third party manager, talent di- rector, mail-order bride agent, prostitution tour agent (Hughes, 1999)
Predator	One who exploits conditions of inequality to buy and/or abuse for personal sexual satisfaction those with less power, e.g. john, punter, buyer, client, customer, trick, pedophile, rapist, sex offender, child molester, pornographer (Hughes, 1999)
Sex Industry	The collection of legal and illegal businesses and single and multi-party operations that profit from the sexual exploitation of women, children, and sometimes, men in trafficking, organized prostitution, and/or pornography; e.g. brothels, massage parlors, bars, strip clubs, mail-order-bride agencies, prostitution tour agencies, "adult enter- tainment," "adult" bookstores, pornographic web sites, etc. (Hughes, 1999)
Sexual Masochist	A person who experiences sexual excitement in the act of being made to suffer.

Exhibit 1.2

Concept	Definition
Sexual Sadist	A person who experiences sexual excitement by inflicting suffering upon another per- son.
Track	A network of cities and other communities regularly traveled to by pimps, traffickers and sexually exploited youth and adults.
Transgender/ Transsexual	A person who is in the process of changing, or who already has changed, his/her natal gender identity to that of the opposite sex. The process involves both hormonal and surgical treatment.
Transvestite	A person who experiences sexual excitement by wearing clothing of the opposite sex.
Trick	Somebody who hires a prostitute (slang); an individual engagement between a pros- titute and a client (slang).
Voyeur	A person who seeks sexual arousal by observing the sexual activity of others.
"White" Slavery	Historically, the concept referred to the abduction and sale of a Caucasian girl or woman into prostitution against her will. Today, the concept does not include a ra- cial designation and, instead, refers to the use of force, deception or other means to compel people into commercial sexual activity.

Organization of the Bibliography

The references contained in this bibliography are divided into six parts: Part I contains references to the very broad field of *Child Rights*; Part II contains references to *Children at Risk*; Part III focuses on references specific to *Child Sexual Exploitation*, including the *commercial* aspects of child sexual exploitation; Part IV contains references dealing with *Laws, the Legal Environment and Legal Resources*; Part V contains references to *International Cooperation in Prosecuting Adults Committing Sexual Crimes Against Children*; and Part VI references relating to the *Use of the Internet in Sexual Crimes Involving Children*. Each part of the bibliography is sub-divided into thematic sections and subsections. In general, every effort has been made to identify empirical studies that report national and international prevalence data concerning child sexual exploitation.

The majority of the references included in the bibliography have been published since 1990, albeit every effort has been made to include seminal works on child sexual abuse, child sexual exploitation, and the commercial sexual exploitation of children published prior to 1990. Similarly, most references are to works published in English and to foreign language publications that have been translated into English.

No doubt many important studies have been overlooked in compiling this listing. The editor sincerely apologies for these omissions. As "a work in progress," readers are encouraged to submit recommendations for additional references to be included in future editions of the bibliography to the author at the following address: restes@ssw.upenn.edu.

Richard J. Estes, D.S.W., Professor and Chair, Concentration in Social and Economic Development Principal Investigator, "The Commercial Sexual Exploitation of Children in the United States, Canada and Mexico" Project (NIJ)

> University of Pennsylvania School of Social Work 3701 Locust Walk/D-15 Philadelphia Pennsylvania 19104-6214 Telephone: 215/898-5531; Fax: 215/573-2099 August, 2001

THE SEXUAL EXPLOITATION OF CHILDREN

A Working Guide to the Empirical Literature

Contents

Preface

PART I CHILD RIGHTS

- A. Rights of the Child
- **B.** Socioeconomic Threats to the Rights of Children 1. Poverty and the Exploitation of Children

PART II CHILDREN AT RISK

A. Child Abuse

- 1. Prevalence
 - 2. Consequences/Correlates of Child Abuse
- 3. Abuse and the Cycle of Violence
- 4. Institutional Responses to the Cycle of Child Domestic Violence

B. Child Sexual Abuse

- 1. Prevalence
- 2. Consequences/Correlates of Child Sexual Abuse

C. Missing and Abducted Children and Youth

- 1. Prevalence
- 2. Recovery of Abducted Children and Youth

D. Runaway, "Throwaway" and Street Children in the United States

- 1. Prevalence
- 2. Causes/Risk Factors Associated With Running Away
- 3. Social and Health Risks of Runaway & Street Youth--Including Sexually Exploited Youth
- 4. Gangs and Gang Culture Among Runaway/Street Youth
- 5. Homeless Youth

PART III CHILD SEXUAL EXPLOITATION

A. Concepts and Definitions

- 1. Commercial Sexual Exploitation of Children (CSEC)
- B. Sexual Predators: Pedophiles, Pimps, Sexual Tourists, and Opportunistic Sexual Exploiters of Children
- C. Prevalence

- D. Impact of Sexual Exploitation on Children
- E. Impact of Sexual Exploitation on Adult Survivors
- F. Forms of Child Sexual Exploitation
 - 1. Pornography
 - 2. Prostitution
 - a) Special Issues Related to Male Prostitution
 - b) Special Issues Related to Female Prostitution
 - c) Special Issues Related to Prostitution and Gay, Lesbian, Bisexual and Transgender Youth
 - d) Child Prostitution in Historical Perspective
 - e) Child Prostitution in International Perspective
 - 3. Trafficking in Children for Sex
 - 4. Child Sex Tourism
 - 5. The U.S. Military and Child Sexual Exploitation

PART IV. USE OF THE INTERNET IN SEX CRIMES INVOLVING CHILDREN

PART V. LAWS, THE LEGAL ENVIRONMENT, AND LEGAL RESOURCES RELATED TO THE SEXUAL EXPLOITATION OF CHILDREN

PART VI. COOPERATION IN PROSECUTING ADULTS COMMITTING SEXUAL CRIMES AGAINST CHILDREN

- 1. Cooperation Between U.S. Public and Private Organizations
- 2. International Cooperation

PART I Child Rights

Section A. Rights of the Child

Brooks, C. et al. 1999. <u>Children's Rights in the</u> <u>United States: In Search of a National Policy</u> (Thousand Oaks CA: Sage Publications).

Cantwell, N. 1992. The origins, developments and significance of the UN Convention on the Rights of the Child, in Detrick, S. <u>The UN Convention on the Rights of the Child: A Guide to</u> <u>the Travaux Preparatoires</u> (Dordrecht: Nijhoff Publishers), p. 22ff.

Korr, Wynne S., Barry J. Fallon and Donald Brieland. 1994. "The UN Convention on the Right of the Child: Implications for social work education," <u>International Social Work</u>, 37(4):333-346.

Levesque, Roger J. R. 1994. Sexual use, abuse and exploitation of children: Challenges in implementing children's rights, <u>Brooklyn Law Review</u> (Fall), pp. 959 ff.

United Nations. 1989. <u>Convention on the Rights</u> of the Child, November 20, 1989. G.A. Res. 44/25, U.N. GAOR, 44th Session, Supplement No. 49, UN Doc. A/44/736 (1989), 28 I.L.M. 1448 (1989).

United Nations Commission on Human Rights (UNCHR). 1992. <u>Rights of the Child: Sale of</u> <u>Children, Child Prostitution, and Child Pornog-</u> <u>raphy</u> (New York: United Nations), E/CN.4/1994/84.

Vittachi, Anuradha. 1989. <u>Stolen Childhood: In</u> <u>Search of the Rights of the Child</u> (New York: Polity Press).

Section B. Socioeconomic Threats to the Rights of Children

Bales, Kevin. 1999. <u>Disposable People: New</u> <u>Slavery in the Global Economy</u> (Berkeley: University of California Press).

Barker, Gary & J. S. Musick. 1994. Rebuilding Nests of Survival: A Comparative Analysis of the Needs of At-risk Adolescent Women and Adolescent Mothers in the U.S., Latin America, Asia, and Africa, <u>Childhood</u> 2(3):152-163.

Bunch, Charlotte. 1997. The intolerable status quo: Violence against women and children, in Unicef, <u>The Progress of Nations</u> (http://www.unicef.org/pon97/women1.htm).

Children's Defense Fund. 2000. <u>The State of</u> <u>America's Children - Yearbook 2000</u>. (Washington, DC: CDF).

Editors. 1996. The flourishing business of slavery, <u>The Economist</u> 340(7984):43-44.

Estes, Richard J. 1995. Social development trends in Africa: The need for a new development paradigm, <u>Social Development Issues</u>, 17(1):18-47.

Estes, Richard J. 1996. Social development trends in Latin America, 1970-1994: In the shadows of the 21st century, <u>Social Development Issues</u>, 18(1):25-52.

Estes, Richard J. 1996. Social development trends in Asia, 1970-1994: The challenges of a new century," <u>Social Indicators Research</u>, 37(2):119-148.

Estes, Richard J. 1997. Trends in European social development, 1970-1994, <u>Social Indicators</u> <u>Research</u>, 38(6):1-19.

Estes, Richard J. 1998a. Social development trends in the successor states to the former Soviet Union: The Search for a New Paradigm, <u>Economies in Transition</u> (Nagoya: United Nations Centre for Regional Development). Estes, Richard J. 1998b. Trends in world social development, 1970-95: Development prospects for a new century, <u>Journal of Developing Societies</u> 14(1):1-29.

Estes, Richard J. and Neil A. Weiner (Editors). 2001. <u>The Silent Emergency: The Sexual Ex-</u> <u>ploitation of Children in the United States, Can-</u> <u>ada and Mexico</u> (Westport CT: Greenwood Publishing Group)—in preparation.

Hood-Brown, Marcia. 1998. Trading for a place: Poor women and prostitution, <u>J of Poverty</u> 2(3):13-33.

Kirschenbaum, Jill. 1995. More cuts to kids, <u>City Limits</u>, XX(5):4.

Knight, Danielle. 1998. Labour Rights: Latin American children exploited on US farms (January 21), [URL: http://oneworld.org/ips2/jan98/usa.html].

Lederer, Laura and Richard Delgado (Editors). 1995. <u>The Price We Pay: The Case Against Racist Speech, Hate Propaganda, and Pornography</u> (New York: Hill and Wang).

Lederer, Laura (Editor). 1998. <u>Speech, Equality</u> and Harm: <u>New Legal Paradigms</u> (Boulder: Westview).

Lie, John. 1996. Globalization and its discontents. <u>Contemporary Sociology</u>, 25(5):585-587.

Rodriquez, Rey. 1998. Rights-LATAM: Globalisation exacerbates children's social ills, (July 6), [URL:

http://oneworld.org/ips2/jul98/18_55_061.html].

United Nations Children's Fund (Unicef). 2000. <u>The State of the World's Children, 2000</u> (New York: Oxford University Press).

U.S. Department of Labor. <u>By the Sweat and</u> <u>Toil of Children</u> [5 Volumes]. (Washington: Department of Labor).

Volume 1: <u>The Use of Child Labor in U.S.</u> <u>Manufactured and Mined Im-</u> <u>ports</u> (1994)

1
)
<u>s</u>
-
-
r

1. <u>Poverty & the Exploitation of</u> <u>Children</u>

Chafel, J. 1993. <u>Child Poverty and Public Pol-</u> icy. (Washington: The Urban Institute).

Lederer, Laura. 1996. Poor children targets of sexual exploitation, <u>National Catholic Reporter</u> 33(5):11-12.

Longford, Michael. 1995. Family poverty and the exploitation of child labor, <u>Law & Policy</u> 17(4): 471-482.

Packard Foundation. 1997. <u>Children in Poverty:</u> <u>Executive Summary.</u> (URL: <u>http://www.futureofchildren.org/cap/exsum_17.</u> <u>htm).</u>

Smeeding, Timothy. 1997. Financial poverty in developed countries: The evidence from LIS. <u>Luxembourg Income Study Working Paper</u> <u>#155</u>, Center for Policy Research, the Maxwell School, Syracuse University.

PART II Children at Risk

Section A *Child Abuse*

Barnett, D., J.T. Manly, and D. Cicchetti. 1993. Defining child maltreatment: The interface between policy and research, in D. Cicchetti and S.L. Toth (eds.), <u>Child Abuse, Child Develop-</u> <u>ment, and Social Policy</u>. (Norwood, NJ: Ablex).

Briere J. et al. 1996. <u>APSAC Handbook on</u> <u>Child Maltreatment</u> (Newbury Park: Sage).

Carter, Lucy S., Lois A. Weithorn and Richard E. Behrman. 1999. Domestic violence and children: Analysis and recommendations, <u>Domestic Violence and Children</u> 9(3):4-40.

Finkelhor, David and J. Dziuba-Leatherman. 1994. Children as victims of violence: A national survey, <u>Pediatrics</u> 94:413-420.

Gelles, Richard J. 1996. <u>The Book of David:</u> <u>How Preserving Families Can Cost Children's</u> <u>Lives</u> (New York: Basic Books).

Gelles, Richard J. 1997 <u>Intimate Violence in</u> <u>Families</u>, 3rd Edition (Newbury Park, Ca: Sage).

Gelles, Richard J. 1998. Untreatable families, in Reece, R.M. (ed.), <u>The Treatment of Child</u> <u>Abuse</u>. (Baltimore, MD: The Johns Hopkins University Press

Marx, S. & P. Toth (Editors). 1993. <u>Investiga-</u> <u>tion and Prosecution of Child Abuse</u>, 2nd Edition. (American Prosecutors Research Institute, National Center for Prosecution of Child Abuse.).

Myers, J.E.B. 1992. <u>Legal Issues in Child</u> <u>Abuse and Neglect</u> (Newbury Park CA: Sage).

National Institute of Justice (NIJ). 1993. Joint Investigations of Child Abuse. (Washington: USGPO). NCCAN/NCPCA. <u>Child Abuse and Neglect</u> <u>State Statue Series</u> (Five Volumes). Prepared by the National Center on Child Abuse and Neglect & the National Center for Prosecution of Child Abuse.

Osofsky, Joy D. 1999. The impact of violence on children, <u>Domestic Violence and Children</u> 9(3):33-49.

Powers, Jane L. & Barbara Jaklitsch. 1989. <u>Un-</u> derstanding Survivors of Abuse: Stories of <u>Homeless and Runaway Adolescents</u> (New York: The Free Press).

Remy, Jacqueline. 1996. Putting abuse in the spotlight: International pressure can help, <u>World</u> <u>Press Review</u> 43(11):7-8.

Schwartz, I.M. & Au Claire, P. (Eds.). 1995. <u>Home-Based Services For Troubled Children</u>. (Lincoln, NE: University of Nebraska Press).

Schwartz, I.M., Rendon, J.A., & Hsieh, C. 1994. Is child maltreatment a leading cause of delinquency? <u>Child Welfare</u>, 73(5), 639-655.

U.S. Department of Justice. 1997. <u>Law En-</u> forcement Response to Child Abuse (Washington: Office of Justice Programs-OJJDP).

Whitcomb, Debra. 1992. <u>When the Victim is a</u> <u>Child.</u> 2nd Edition. (Washington: Department of Justice, Office of Justice Programs).

1. <u>Prevalence</u>

Cosner, R.E., Weiner, N.A., Huang, V. et al. 1997. <u>Knowingly False and Malicious Reporting</u> of Child Abuse and Neglect in Pennsylvania: <u>Critical Questions, Findings, and Recommendations</u>. (Harrisburg, PA: Pennsylvania Department of Public Welfare: Office of Children, Youth, and Families).

Editors. 1996. Disturbing new figures about child abuse, <u>Maclean's</u> 110(29):15.

Fantuzzo, John W. and Wanda K. Mohr. 1999. Prevalence and effects of child exposure to domestic violence, <u>Domestic Violence and Chil-</u> <u>dren</u> 9(3):21-32.

Jones, Lisa & David Finkelhor. 2001. The decline in child sexual abuse cases, <u>OJJDP Bulle-</u> <u>tin</u> (January).

Kilpatrick, Dean & Benjamin Saunders. 1997. The prevalence and consequences of child victimization, <u>Research Preview</u> (of the NIJ), April, pp. 1-4.

U.S. Dept. of Health and Human Services. 2000. Child Maltreatment 1998: Reports From the States to the National Child Abuse and Neglect Data System (Washington: U.S. Government Printing Office).

2. <u>Consequences/Correlates of Child</u> <u>Abuse</u>

Kelley, Barbara, Thornberry, Terence & Smith, C. 1997. In the wake of child maltreatment, <u>Ju-</u><u>venile Justice Bulletin</u> (August), pp. 1-15.

Silverman, Amy, Helen Z. Reinherz & Rose M. Giaconia. 1996. The long-term sequelae of child and adolescent abuse: A longitudinal community study, <u>Child Abuse and Neglect</u> 20(8):709-723.

Ireland, T., and Widom, Cathy S. 1994. Childhood victimization and risk for alcohol and drug arrests. <u>The International J of the Addictions</u> 2(2):235–274.

Smith, C., and T.P. Thornberry. 1995. The relationship between childhood maltreatment and adolescent involvement in delinquency. <u>Criminology</u> 33(4):451–481.

3. Abuse and the Cycle of Violence

Reichert, Jennifer L. 1998. Most male felons victimized as children, study says, <u>Trial</u> 34(7):120-121.

Straus, M.A. & Gelles, Richard J. 1988. <u>Intimate</u> <u>Violence</u>. (New York, NY: Simon and Schuster).

Weiner, Neil A. & Wolfgang, M. 1989. "Violent Criminal Careers and Violent Career Criminals': An Overview of the Research Literature," in Neil A. Weiner and Marvin E. Wolfgang (Editors). <u>Violent Crime, Violent Criminals</u>. (Newbury Park, CA: Sage Publications), pp. 35-138.

Weiner, Neil A. 1994. "The Probability and Timing of Rearrest for Serious Violent Crimes: Some Descriptive Patterns in Individual Arrest Histories and Their Policy Implications," in H. J. Kerner and E. Weitekamp (Editors). Cross and National Longitudinal Research on Human Development and Criminal Behavior. (The Netherlands: Kluwer Publishers).

Weiner, Neil A. & R. Barry Ruback. 1995. <u>Violent Interpersonal Behavior Social and</u> <u>Cultural Aspects.</u> (New York: Springer).

Widom, Cathy S. 1996. The cycle of violence revisited, <u>Research Preview (of the NIJ)</u>. Washington, DC: National Institute of Justice.

4. <u>Institutional Responses to the Cy-</u> cle of Child Domestic Violence

Culross, Patti L. 1999. Health care system response to children exposed to domestic violence, <u>Domestic Violence and Children</u> 9(3):111-121.

Findlater, Janet E. and Susan Kelly. 1999. Child protective services and domestic violence, <u>Do-</u> <u>mestic Violence and Children</u> 9(3):84-96.

Golden, Olivia. 2000. The federal response to child abuse and neglect, <u>American Psychologist</u> 55(9):1050-1053.

Groves, Betsy McAlister. 1999. Mental health services for children who witness domestic violence, <u>Domestic Violence and Children</u> 9(3):122-132. Lemon, Nancy K.D. 1999. The legal system's response to children exposed to domestic violence, <u>Domestic Violence and Children</u> 9(3):67-83.

Matthews, Martha A. 1999. The impact of federal and state laws on children exposed to domestic violence, <u>Domestic Violence and Children</u> 9(3):50-66.

Saathoff, Amy J. and Elizabeth A. Stoffel. 1999. Community-based domestic violence services, <u>Domestic Violence and Children</u> 9(3):97-110.

Wolfe, David A. and Peter G. Jaffee. 1999. Emerging strategies in the prevention of domestic violence, <u>Domestic Violence and Children</u> 9(3):133-144.

Section B. Child Sexual Abuse

Bala, N. & Schwartz, I.M. 1993. Legal responses to the juvenile sexual offender, in H.E. Barbaree, W.L. Marshall, & D.R. Laws (Eds.), <u>The Juvenile Sexual Offender</u> (New York: Guilford Publications, Inc.).

Bienen, Leigh B. 1998. <u>Defining Incest</u>, Special issue of the <u>Northwestern University Law Review</u> 92(4).

Bryan, J. 1987. <u>Team Investigation in Child</u> <u>Sexual Abuse Cases: A Desk Reference for Law</u> <u>Enforcement Officers, Protective Service Work-</u> <u>ers, and Prosecuting Attorneys</u> (Little Rock, AK: Arkansas Child Sexual Abuse Commission).

Center for the Future of Children. 1994. <u>Sexual</u> <u>Abuse of Children</u>. (Newbury Park CA: Sage).

Cole, Joy. 1993. The child as victim of extrafamilial sexual exploitation with reference to the characteristics and modus operandi of the offender: A victimological study. <u>Dissertation</u> <u>Abstracts International</u> 53(11): 4091-A-4092-A.

De Mause, Lloyd. 1991. The universality of incest, J of Psychohistory, 19(2):123-164.

Finkelhor, D. et al. 1986. <u>A Sourcebook on</u> <u>Child Sexual Abuse</u> (Newbury Park CA: Sage).

Goldman, R. & V. Wheeler. 1986. <u>Silent Shame:</u> <u>The Sexual Abuse of Children and Youth</u> (Danville IL: The Interstate).

Haugaard, Jeffrey J. 2000. The challenge of defining child sexual abuse, <u>American Psy-chologist</u> 55(9):1036-1039.

Knight, Raymond, D.L. Carter, and Robert A. Prentky. 1989. A System for the Classification of Child Molesters: Reliability and Application," J of Interpersonal Violence 4:3–23.

Lanning, Kenneth V. 1992. <u>Child Molesters: A</u> <u>Behavioral Analysis for Law Enforcement Offi-</u> <u>cers Investigating Cases of Child Sexual Exploi-</u> <u>tation</u>. 3rd Edition. (Arlington VA: National Center for Missing and Exploited Children).

Lanning, Kenneth V. 1992. <u>Child Sex Rings: A</u> <u>Behavioral Analysis</u> (Arlington VA: National Center for Missing and Exploited Children).

National Center for Missing and Exploited Children. <u>Child Molesters Who Abduct: Summary of</u> <u>the Case in Point Series</u>, (Arlington VA: NCMEC), 116 pp.

Pence, D. & C. Wilson. 1994. <u>Team Investiga-</u> tion of Child Sexual Abuse (Newbury Park CA: Sage).

Prentky, Robert, Knight, Raymond & Lee, Austin. 1997. <u>Child Sexual Molestation: Research</u> <u>Issues</u>. (Washington: National Institute of Justice).

Rickel, Annette U. & Marie C. Hendren. 1993. Aberrant sexual experiences, in Gullotta, Thomas P., Gerald R. Adams & Raymond Montemayor (Editors), <u>Adolescent Sexuality</u> (Thousand Oaks CA: Sage Publications).

Saywitz, Karen et al. 2000. Treatment for sexually abused children and adolescents, <u>American</u> <u>Psychologist</u> 55(9):1040-1049.

Snyder, Howard N. 2000. <u>Sexual Assault of</u> <u>Young Children as Reported to Law Enforce-</u> <u>ment: Victim, Incident, and Offender Character-</u> <u>istics</u> (Washington: U.S. Department of Justice, Office of Justice Programs).

U.S. Department of Justice. 1997. <u>Interviewing</u> <u>Child Witnesses and Victims of Sexual Abuse</u> (Washington: Office of Justice Programs--OJJDP).

U.S. Department of Justice. 1997. <u>Criminal In-</u> vestigation of Child Sexual Abuse (Washington: Office of Justice Programs---OJJDP).

Wycoff, M.A. & M. Kealoha. 1987. <u>Creating the</u> <u>Multidisciplinary Response to Child Sex Abuse:</u> <u>An Implementation Guide</u> (Washington: Police Foundation).

1. <u>Prevalence</u>

Feldman, William et al. 1991. Is childhood sexual abuse really increasing in prevalence? An analysis of the evidence, <u>Pediatrics</u> 88(1):29-34.

Gelles, Richard J. & Wolfner, G.D. 1994. Sexual offending and victimization: A life course perspective. In Rossi, A.S. (Ed.), <u>Sexuality</u> <u>Across the Life Course</u>. (Chicago, IL: University of Chicago Press).

Greenfeld, Lawrence. 1997. <u>Sex Offenses and</u> <u>Offenders: An Analysis of Data on Rape and</u> <u>Sexual Assault.</u> (Washington: Dept. of Justice, Bureau of Justice Statistics), NCJ-163392.

Kilpatrick, Dean and Benjamin Saunders. 1997. The Prevalence and Consequence of Child Victimization, NIJ Research Preview, http://www.ncjrs.org.

Pandora's Box. 1999. Sexual abuse statistics, http://pages.prodigy.com/faulkner/stats.htm.

Terry, Sara. 1991. Sins of the innocent, <u>Rolling</u> <u>Stone</u> (October 31), pp. 67-71.

2. <u>Consequences/Correlates of Child</u> <u>Sexual Abuse</u>

Arata, C.M. 2000. From child victim to adult victim: A model for predicting sexual re-victimization, <u>Child Maltreatment</u> 5:28-38.

Briere, J. 1988. The long-term clinical correlates of childhood sexual victimization, <u>Annals of the</u> <u>New York Academy of Sciences</u> 528:327-334.

Briere, J. & M. Runtz. 1988. Symptomology associated with childhood sexual victimization in a non-clinical adult sample, <u>Child Abuse and</u> <u>Neglect</u> 12(1):55-59.

Browne, A. & D. Finkelhor. 1986. Impact of sexual abuse: A review of the research, <u>Psychological Bulletin</u> pp. 66-77.

Collins, Mary E. 2001. The potential for later revictimization among child victims: A review of research and implications for practice and policy, unpublished paper presented at the July 2001 NIJ Criminal Justice Research and Evaluation Conference.

Dembo, R., L. Williams, W. Wothke, J. Schmeidler, and C.H. Brown. 1992. The role of family factors, physical abuse, and sexual victimization experiences in high-risk youth's alcohol and other drug use and delinquency: A longitudinal model, <u>Violence and Victims</u> 7(3): 245-266.

Finkelhor, D. and A. Brown. 1985. The traumatic impact of child sexual abuse: A conceptualization, <u>American J of Orthopsychiatry</u> 55(October).

Irwin, H.J. 1999. Violent and non-violent revictimization of women abused in childhood, <u>Jour-</u> <u>nal of Interpersonal Violence</u> 14:1095-1110.

Kessler, B.L. & K.J. Bieschke. 1999. A retrospective analysis of shame, dissociation, and adult victimization in survivors of childhood sexual abuse, <u>Journal of Counseling Psychology</u> 46:335-341. Mayall, A. & S.R. Gold. 1995. Definitional issues and mediating variables in the sexual revictimization of women sexually abused as children, <u>Journal of Interpersonal Violence</u> 10:26-42.

McClanahan, S.F. et al. 1999. Pathways into prostitution among female jail detainees and their implications for mental health services, <u>Psychiatric Services</u> 50(12):1606-13.

Molnar, Beth E. et al. 1998. Suicidal behavior and sexual/physical abuse among street youth, <u>Child Abuse and Neglect</u> 22(3):213-222.

Mullen, P. E. et al. 1996. The long-term impact of the physical, emotional, and sexual abuse of children: A community study, <u>Child Abuse and</u> <u>Neglect</u> 20(1):7-20.

National Center for Missing and Exploited Children. <u>Children Traumatized in Sex Rings</u>, (Arlington VA: NCMEC), 56 pp.

Potter, K. et al. 1999. Early developmental experiences of female sex workers: A comparative study, <u>Australian and New Zealand J of Psychia-</u> try 33(6):935-40.

Prentky, Robert and Raymond A. Knight. 1993. Age of Onset of Sexual Assault: Criminal and Life History Correlates," in <u>Sexual Aggression:</u> <u>Issues in Etiology, Assessment, and Treatment,</u> eds. G.C.N. Hall, R. Hirschman, J.R. Graham, and M.S. Zaragoza (Washington, DC: Taylor & Francis), pp. 43–62.

Rotheram-Borus, Mary Jane et al. 1996. Sexual abuse history and associated multiple risk behavior in adolescent runaways, <u>American J of</u> <u>Orthopsychiatry</u> 66(3):390-400.

Schafer, M. et al. 2000. Sexual and physical abuse during early childhood or adolescence and later drug addiction, <u>Psychotherapie</u>, <u>Psychoso-</u> <u>matik</u>, <u>Medizinische Psychologie</u> 50(2):38-50.

Seng, Magnus J. 1989. Child sexual abuse and adolescent prostitution: A comparative analysis, <u>Adolescence</u> 24(95):665-675.

Silbert, Mimi H. 1982. Prostitution and sexual assault: Summary of results, <u>International Journal of Biosocial Research</u> 3(2):69-71.

Silbert, Mimi H., Ayala M. Pines. 1981. Sexual child abuse as an antecedent to prostitution, <u>Child Abuse & Neglect 5(4):407-411.</u>

Silbert, Mimi H., Ayala M. Pines. 1983. Early sexual exploitation as an influence in prostitution, <u>Social Work</u> 28(4):285-289.

Simons, Ronald & Les B. Whitbeck. 1991. Sexual abuse as a precursor to prostitution and victimization among adolescent and adult homeless women, <u>J of Family Issues</u> 12:361-379.

Widom, Cathy S. & M. Ashely Ames. 1994. Criminal consequences of childhood sexual victimization, <u>Child Abuse and Neglect</u> 18(4):303-318.

Widom, Cathy S. & Joseph B. Kuhns. 1996. Childhood victimization and subsequent risk for promiscuity, prostitution and teenage pregnancy: A prospective study, <u>American J of Public</u> <u>Health</u> 86(11):1607-1612.

Section C. Missing and Abducted Children and Youth

National Center for Missing and Exploited Children. <u>An Analysis of Infant Abductions</u>, (Arlington VA: NCMEC), 62 pp.

National Center for Missing and Exploited Children. <u>Family Abduction</u>, (Arlington VA: NCMEC).

1. <u>Prevalence</u>

Finkelhor, David. 1990. <u>National Incidence</u> <u>Studies on Missing, Abducted, Runaway, and</u> <u>Thrownaway Children in America</u> (Washington: U.S. Department of Justice, Office of Juvenile Justice and Delinquency Prevention). Hansen, Louise. 2000. <u>Second Comprehensive</u> <u>Study of Missing Children.</u> (Washington: U.S. Department of Justice, Office of Justice Programs, Office of Juvenile Justice and Delinquency Prevention).

2. <u>Recovery of Missing and Exploited</u> <u>Children</u>

National Center for Missing and Exploited Children. <u>Recovery and Reunification of Missing</u> <u>Children: A Team Approach</u>, (Arlington VA: NCMEC), 52 pp.

Section D. Runaway, "Throwaway" and Street Children in the United States

Bass, Deborah. 1992. <u>Helping Vulnerable</u> <u>Youths: Runaway and Homeless Adolescents in</u> <u>the United States</u> (Washington DC: National Association of Social Workers Press).

Coates, R. 1990. <u>A Street is Not a Home: Solv-</u> ing America's Homeless Dilemma (Buffalo: Promethus Books).

Ennett, S.T. et al. 1999. Social network characteristics associated with risky behaviors among runaway and homeless youth, <u>J of Health and</u> <u>Social Behavior</u> 40(1):63-78.

Hecht, Tobias. 1998. <u>At Home in the Street:</u> <u>Street Children of Northeast Brazil</u> (Cambridge: Cambridge University Press).

Hersch, P. 1988. Coming of age on city streets, <u>Psychology Today</u> (January):28-36.

Hoffman, D. and N. Scheper-Hughes. 1997. On the streets of America, <u>Natural History</u> 106(6):44-53.

Kufeldt, Kathleen & Margaret Nimmo. 1987. Youth on the street: Abuse and neglect in the eighties, <u>Child Abuse and Neglect</u> 11(4):531-543.

Le Roux, J. 1996. The worldwide phenomenon of street children: Conceptual analysis, <u>Adoles-cence</u> 31(Winter):965-971.

Le Roux, J. & C. Smith. 1998. Causes and characteristics of the street children phenomenon: A global perspective, <u>Adolescence</u> 33(131):683-688.

Le Roux, J. and C. Smith. 1998. Public perceptions of, and reactions to, street children, <u>Adolescence</u> 33(132):901-913.

Lucas, B. M. & L. Hackettt. 1995. <u>Street Youth:</u> <u>On Their Own in Indianapolis</u> (Indianapolis IN: Health Foundation of Greater Indianapolis).

Paradise & Horowitz. 1994. <u>Runaway and</u> <u>Homeless Youth: A Survey of State Law</u> (Washington DC: American Bar Association, Center on Children and the Law).

Rothman, Jack. 1991. <u>Runaway and Homeless</u> <u>Youth</u> (White Plains NY: Longman Publishing Group).

Schaffner, L. 1998. Search for connection: A new look at teenaged runaways, <u>Adolescence</u> 33(131):619-27.

Smollar, J. 1999. Homeless youth in the United States: Description and developmental issues, <u>New Directions for Child and Adolescent De-</u> <u>velopment</u> 85:47-58.

Tirney, N. 1997. <u>Robbed of Humanity: Lives of</u> <u>Guatemalan Street Children</u> (Canada: Pangaea).

Wurzbacher, Kathryn V., E. D. Evans, & E. J. Moore. 1991. Effects of Alternative Street School on youth involved in prostitution, <u>Adolescent Health</u> 12(7):549-554.

Yates, Gary L. et al. 1991. A risk profile comparison of homeless youth involved in prostitution and homeless youth not involved, <u>J of Adolescent Health</u> 12(7):545-548. Yates, Gary L. et al. 1988. A risk profile comparison of runaway and non-runaway youth, <u>American J of Public Health</u> 78(7):820-821.

Youth Plus. 1995. <u>Working With Street Children: Selected Case Studies from Africa, Asia, and Latin America</u>. (Paris: Unesco Publishing for the International Catholic Child Bureau).

Zide, Marilynn R & Andrew L. Cherry. 1992. A typology of runaway youths: An empirically based definition, <u>Child and Adolescent Social</u> <u>Work J</u> 9(2):155-168.

1. <u>Prevalence</u>

Finkelhor. 1990. <u>National Incidence Studies on</u> <u>Missing, Abducted, Runaway, and Thrownaway</u> <u>Children in America</u> (Washington: U.S. Department of Justice, Office of Juvenile Justice and Delinquency Prevention).

McCormack, Arlene, Mark-David Janus & Ann W. Burgess. 1986. Runaway youths and sexual victimization: Gender differences in an adolescent runaway population, <u>Child Abuse and Neglect</u> 10(3):387-395.

2. <u>Causes/Risk Factors Associated</u> With Running Away

Ayerst, S.L. 1999. Depression and stress in street youth, <u>Adolescence</u> 34(135):567-75.

Booth, R.E. et al. 1999. The challenge of changing drug and sex risk behaviors of runaway and homeless adolescents, <u>Child Abuse and Neglect</u> 23(12):1295-306.

DiPaolo, Michael. 1999. <u>The Impact of Multiple</u> <u>Childhood Trauma on Homeless Runaway Ado-</u> <u>lescents</u> (New York: Garland Publishing, Inc.).

Farber, E.D. et al. 1984. Violence in families of adolescent runaways, <u>Child Abuse & Neglect</u> 8:295-299.

Kaufman, Jeanne G. & Cathy S. Widom. 1999. Childhood victimization, running away and delinquency, <u>Journal of Research in Crime and</u> <u>Delinquency</u> 36(4):347-370.

Powers, Jane L. & Barbara Jaklitsch. 1989. <u>Un-</u> derstanding Survivors of Abuse: Stories of <u>Homeless and Runaway Adolescents</u> (New York: The Free Press).

Ryan, K.D. et al. 2000. Psychological consequences of child maltreatment in homeless adolescents: Untangling the unique effects of maltreatment and family environment, <u>Child Abuse</u> and <u>Neglect</u> 24:333-352.

Smith, N. 1991. <u>Confronting Sexual Exploita-</u> tion of Homeless Youth: California's Juvenile <u>Prostitution Intervention Projects</u> (Sacramento CA: California Office of Criminal Justice Planning).

Stiffman, Arlene R. et al. 1989. Physical and sexual abuse in runaway youths, <u>Child Abuse</u> and <u>Neglect</u> 13(3):417-426.

3. <u>Social and Health Risks of Run-</u> away & Street Youth--Including <u>Sexually Exploited Youth</u>

Bond, Lydia S, Rafael Mazin, and Maria V. Jiminez. 1992. Street youth and AIDS, <u>AIDS</u> <u>Education and Prevention</u> 4(3):14-23.

Buckner, J.C. 1998. Displaced children: Meeting the health, mental health, and educational needs of immigrant, migrant and homeless youth, <u>Adolescent Medicine</u> 9(2):323-34.

Freese, T. E. & R. G. MacKenzie. 1992. Highrisk behavior among male street youth in Hollywood CA, <u>AIDS Education and Prevention</u> (supplement) (Fall):24-33.

Greene, J.M. et al. 1999. Prevalence and correlates of survival sex among runaway and homeless youth, <u>American J of Public Health</u> 89(9):1406-9. Greene, J.M. and C. L. Ringwalt. 1998. Pregnancy among three national samples of runaway and homeless youth, <u>J of Adolescent Health</u> 23(6):370-7.

Hermann, R.C. 1988. Center provides approach to major social ill: Homeless urban runaways, "throwaways," <u>J of the American Medical Association</u> 260(3):311-312.

Johnson, Timothy P. et al.. 1996. Self reported risk factors for AIDS among homeless youth, <u>AIDS Education and Prevention</u> 8(4):308-322.

Kipke, M.D. et al. 1997. "Substance abuse" disorders among runaway and homeless youth, <u>Substance Use and Misuse</u> 32(7-8):969-86.

Kipke, M.D. et al. 1997. Homeless youth and their exposure to and involvement in violence while living on the streets, <u>J of Adolescent</u> <u>Health</u> 20(5):360-7

Kral, Alex H. et al. 1997. Prevalence of sexual risk behavior and substance use among runaway and homeless adolescents in San Francisco, Denver and New York City, <u>International J of</u> <u>STD & AIDS</u> 109 ff.

McCarthy, Bill & John Hagan. 1992. Surviving on the street: The experiences of homeless youth, <u>J of Adolescent Research</u> 7(4):412-430.

Molnar, Beth E. et al. 1998. Suicidal behavior and sexual/physical abuse among street youth, <u>Child Abuse & Neglect</u> 22(3):213-222.

Morey, M. A. & L. S. Friedman. 1993. Health care needs of homeless adolescents, <u>Current</u> <u>Opinion in Pediatrics</u> 5(4):395-399.

Pennbridge, Julia N. et al. 1992. High risk behaviors among male street youth in Hollywood CA <u>AIDS Education and Prevention</u>, Supplement, (Fall):24-33.

Ringwalt, C.L. et al. 1998. Familial backgrounds and risk behaviors of youth with thrownaway experiences, <u>J of Adolescence</u> 21(3):241-52. Rotheram-Borus, Mary Jane, Cheryl Koopman & A. A. Ehrhardt. 1991. Homeless youth and HIV infection, <u>American Psychologist</u> 46(11):1188-1197.

Sherman, D.J.. 1992. The neglected health care needs of street youth, <u>Public Health Reports</u> 107(4):433-440.

Silbert, Mimi H., Ayala M. Pines. 1981b. Occupational hazards of street prostitutes, <u>Criminal</u> <u>Justice & Behavior</u> 8(4):395-399.

Silbert, Mimi H., Ayala M. Pines. 1982. Victimization of street prostitutes, <u>Victimology</u> 7(1-4):122-133.

Silbert, Mimi H., Ayala M. Pines, Teri Lynch. 1982. Substance abuse and prostitution, <u>Journal</u> <u>of Psychoactive Drugs</u>. 14(3):193-197.

Sullivan, Richard T. 1996. The challenge of HIV prevention among high-risk adolescents, <u>Health and Social Work</u> 21(1):58-65.

Unger, Jennifer B. et al. 1998. Stress, coping, and social support among homeless youth, <u>J of</u> <u>Adolescent Research</u> 13(2):134-157.

Ward-Russell, L. M. et al. 1992. <u>A Cry for Help:</u> <u>The Mental Health and Service Needs of Home-</u> <u>less Sexually Exploited Youth</u> (Sacramento CA: California Office of Criminal Justice Planning).

Whitbeck, Les B. & Ronald L. Simons. 1990. Life on the streets: The victimization of runaway and homeless adolescents, <u>Youth & Society</u> 22(1):108-125.

Wolf, Susan M. 1990. Health care needs of homeless and runaway youths, <u>J of the American Medical Association</u> (JAMA) 263(6):811-812.

4. <u>Gangs and Gang Culture Among</u> <u>Runaway and Street Youth</u>

Budnick, Kimberly J. & Ellen Shields-Fletcher. 1998. What about the girls? <u>OJJDP Fact Sheet</u> <u>#84</u> [http://www.ncjrs.org/txtfiles/fs9884.txt]. Bureau of Justice Assistance. 1997. <u>Urban Street</u> <u>Gang Enforcement</u>. (Washington: U.S. Department of Justice, Office of Justice Programs, Bureau of Justice Assistance).

Bilchik, Shay. 1999. <u>1996 National Youth Gang</u> <u>Survey: Summary</u> (Washington: Department of Justice, Office of Juvenile Justice and Delinquency Prevention).

Egley, Arlen. 2000. <u>Highlights of the 1999 Na-</u> tional Youth Gang Survey, 1999. (Washington: Department of Justice, Office of Juvenile Justice and Delinquency Prevention).

Hazelhurst, Kayleen & Cameron Hazlehurst (Editors). 1998. <u>Gangs and Youth Subcultures:</u> <u>International Explorations</u> New Brunswick NJ: Transaction Publishers.

Howell, James C. 1997. Youth gangs, <u>OJJDP</u> <u>Fact Sheet #72</u> [http://www.ncjrs.org/txtfiles/fs9772.txt].

Moore, John P. & Craig P. Terrett. 1999. Highlights of the 1997 national youth gang survey, <u>OJJDP Fact Sheet #97</u> [<u>http://www.ncjrs.org/txtfiles1/fs9997.txt.</u>]

Moore, John P. and Ivan L. Cook. 1999. <u>High-lights of National Youth Gang Survey, 1998</u>. (Washington: Department of Justice, Office of Juvenile Justice and Delinquency Prevention).

Puzzanchera, Charles M. 1996. The youngest offenders, 1996, <u>OJJDP Fact Sheet #87</u> [<u>http://www.ncjrs.org/txtfiles1/fs-9887.txt</u>.]

5. <u>Homeless Youth</u>

Cwayna, Kevin. 1993. <u>Knowing Where the</u> <u>Fountains Are: Stories and Stark Realities of</u> <u>Homeless Youth, 1993</u>. (Minneapolis: Fairview Press).

Jarvis, Sara and Robert Robertson. 1993. <u>Transi-</u> tional Living Programs for Homeless Adolescents (Washington: Georgetown University Child Development Center).

National Coalition for the Homeless. 1999a. How many people experience homelessness? <u>NCH Fact Sheet #2</u> (Washington: NCH).

National Coalition for the Homeless. 1999b. Who is homeless? <u>NCH Fact Sheet #3</u> (Washington: NCH).

National Coalition for the Homeless. 1999c. Homeless families with children, <u>NCH Fact</u> <u>Sheet #7</u> (Washington: NCH).

National Coalition for the Homeless. 1999d. Homeless youth, <u>NCH Fact Sheet #11</u> (Washington: NCH).

Pires, Sheila and Judith Silber. 1991. On Their Own: Runaway and Homeless Youth and Programs That Serve Them. (Washington: Georgetown University Child Development Center).

Roman, Nan P. and Phyllis B. Wolfe. 1995. Web of Failure: The Relationship Between Foster Care and Homelessness. (Washington: National Alliance to End Homelessness).

Shinn, Marybeth and Beth Weitzman. 1996. Homeless families are different," in <u>Homeless-ness in America</u>. Washington: National Coalition for the Homeless).

U.S. Conference of Mayors. 1998. <u>A Status Report on Hunger and Homelessness in America's Cities, 1998</u> (Washington: U.S. Conference of Mayors).

U.S. Department of Health and Human Services. 1997. <u>National Evaluation of Runaway and</u> <u>Homeless Youth</u>. (Silver Spring MD: National Clearinghouse on Families and Youth).

Zangrillo, Patricia and Monique Mercer. 1995. Housing and Foster Care: Results of a National Survey (Washington: American Public Welfare Assocation).

PART III Child Sexual Exploitation

Section A. Concepts and Definitions

Bullough, Bonnie & Vern L. Bullough. 1997. Prostitution, <u>Collier's Encyclopedia</u>, p. 97ff.

Cassidy, June. 1996. The unseen menace, <u>Nurs-ing Times</u> 92(8):50-51.

Childley, Joe. 1996. A fight to save children: Activists battle the sexual exploitation of kids, <u>Maclean's</u> 109(36):20-22.

Chung, Janie. 1998. Redirecting the debate over trafficking in women: Definitions, paradigms, and contexts, <u>Harvard Human Rights J</u>(Spring), pp. 65ff.

Coles, R. & G. Stokes. 1985. <u>Sex and the</u> <u>American Teenager</u> (New York: Harper Colophon).

Davidson, Howard and Gregory Loken. 1987. Child Pornography and Prostitution: Background and Analysis. (Arlington, Virginia: National Center for Missing and Exploited Children, pp. 66-68).

Downs, A. C. & L.S. Hillje. 1993. Historical and theoretical perspectives on adolescent sexuality: An overview, in Gullotta, T.P et al. (Editors), <u>Adolescent Sexuality</u> (Newbury Park CA: Sage Publications).

Editors. 1996. Fighting child sex, <u>Maclean's</u> 109(16):45.

Editors. 1993. Prostitution, <u>Columbia Encyclopedia</u>. 5th Edition.

Ennew, Judith. 1986. <u>The Sexual Exploitation of</u> <u>Children</u> (New York: St. Martin's Press).

Ennew, Judith. 1988. <u>The Next Generation:</u> <u>Third World Children and the Future</u> (New Haven: New Society Publishers). Ennew, Judith. 1996. <u>Exploitation of Children</u> (Chatham: Raintree Steck-Vaughan Publishers).

Ennew, Judith et al. 1996. <u>Children and Prostitution: How Can We Measure and Monitor the</u> <u>Commercial Sexual Exploitation of Children?</u> <u>Literature Review and Annotated Bibliography</u> (New York: Unicef).

Finkelhor, D. & J. Dziuba-Leatherman. 1994. Victimization of children, <u>American Psychologist</u> 3:173-183.

Gilfoyle, Timothy J. 1991. Prostitution (in U.S. History), <u>Reader's Companion to American History</u>, p. 875-878.

Goldstein, Seth L. 1987. <u>The Sexual Exploita-</u> tion of Children: A Practical Guide to Assessment, Investigation, and Intervention. (New York: Elsevier).

Holmes, R. A. 1991. <u>Sex Crimes</u> (Newbury Park CA: Sage Publications).

Mahler, Karen. 1996. Global concern for children's rights: The World Congress Against Sexual Exploitation, http://www.agaiusa.org/pubs/journals/2307997.html.

Massachusetts Child Exploitation Network. 1995. <u>Child Sexual Exploitation: Improving In-</u><u>vestigations and Protecting Victims--Blueprint</u> for Action. (Newton MA: Education Development Center, Inc.).

Nandon, Susan M. et al. 1998. Antecedents to prostitution: Childhood victimization, <u>Journal of Interpersonal Violence</u> 13:206ff.

Russell, D. E. 1984. <u>Sexual Exploitation</u> (Beverly Hills CA: Sage Publications).

U.S. Department of Justice. 1997. <u>Understand-ing and Investigation Child Sexual Exploitation</u> (Washington: Office of Justice Programs--OJJDO).

U.S. House Select Committee on Children, Youth, and Families. 1985. <u>Child Victims of</u> Exploitation,. 99th Congress, 1st Session (Washington: GPO).

Weir, I. And M. Wheatcroft. 1995. Allegation of children's involvement in ritual sexual abuse: Clinical experiences of 20 cases, <u>Child Abuse</u> and <u>Neglect</u> 19(4):491-505.

1. The Commercial Sexual Exploitation of Children (CSEC)

Barnitz, Laura A. 1998. <u>Commercial Sexual Exploitation of Children: Youth Involved in Prostitution, Pornography, and Sex Trafficking</u> (Washington: Youth Advocate Program International).

Karlen, Helena & Christina Hagner. 1996. Commercial Sexual Exploitation of Children in Some Eastern European Countries (New York: ECPAT-USA).

Munir, A B & Yasin, S. 1997. Commercial sexual exploitation (of children), <u>Child Abuse Re-</u> <u>view</u> 6(2):147-153.

Muntarbhorn, Vitit. 1996. <u>Report of the Special</u> <u>Rapporteur on the Sale of Children, Child Prostitution, and Child Pornography</u>, United Nations Economic and Social Council, Commission on Human rights, 52nd Session, Agenda Item 20 (New York: UN), E/CN.4/1996/100 (1996).

United Nations. 1994. <u>Promotion and Protection</u> of the Rights of Children: Sale of Children, <u>Child Prostitution and Child Pornography: Note</u> by the Secretary-General, (New York: U.N. General Assembly), Doc. A/49/478, October 5, 1994, p.29.

Youth Advocate Program International (YAPI). 1998. <u>Commercial Sexual Exploitation of Chil-</u> <u>dren: Youth Involved in Prostitution, Pornogra-</u> <u>phy and Sex Trafficking (Washington DC:</u> YAPI).

World Health Organization. 1996. Commercial sexual exploitation of children: The health and

psychosocial dimensions, paper presented at the World Congress Against Commercial Sexual Exploitation of Children, Stockholm, Sweden (August 27-31).

World Vision International. 1996. <u>The Commer-</u> cial Sexual Exploitation of Children.

Section B. Sexual Predators: Pedophiles, Pimps, Sexual Tourists, and Opportunistic Sexual Exploiters of Children

Becker, J.V. et al. 1986. Characteristics of adolescent incest sexual perpetrators: Preliminary findings, Journal of Family Violence 1:85-97.

Briere, John & Marsha Runtz. 1989. University males' sexual interest in children: Predicting potential indices of 'pedophilia' in a non-forensic sample, <u>Child Abuse and Neglect</u> 13(1):65-75.

Dowling, Denise. 2000. "Pimpin" is hard work, <u>Health and Body</u> (January 29).

Faugier, Jean & Mary Sargent. 1997. Boyfriends, "pimps" and clients, in , in Scambler, Graham & Annette Scambler (Editors), <u>Rethinking Prostitution: Purchasing Sex in the 1990s</u> (London: Routledge).

Giobbe, Evelina. 1993. An analysis of individual, institutional, and cultural pimping, <u>Michi-</u> gan J of Gender and Law 33.

Giobbe, Evelina. 1993. A comparison of pimps and batters, <u>Michigan J of Gender and Law</u> 1(1):33-57.

Inciardi, James A. 1984. Little girls and sex: A glimpse into the world of the "baby pro," <u>Deviant Behavior</u> 5(1-4):71-78.

Inciardi, James A. 1993. Chicken heads, slow necks, freaks and blood suckers: A glimpse at the Miami sex-for-crack market, in Ratner, Mitchell S. (Editor) <u>Crack Pipe as Pimp: An</u> Ethnographic Investigation of Sex-For-Crack Exchanges (New York: Free Press).

O'Connell Davidson, Julia. 1996. The sex exploiter, paper presented at the World Congress Against the Sexual Exploitation of Children, Stockholm (August 27-31).

Okami, Paul (1992) Child perpetrators of sexual abuse: The emergence of a problematic deviant category, <u>Journal of Sex Research</u> 29:109-30.

Section C. Prevalence of Child Sexual Exploitation

Federal Bureau of Investigation. 1998. <u>Uniform</u> <u>Crime Reports for the United States</u> (Washington DC: U.S. Department of Justice).

Flowers, R.B. 1994. <u>Victimization and Exploita-</u> tion of Women and Children: A Study of Physical, Mental and Sexual Maltreatment in the <u>United States</u>. (Jefferson NC: McFarland Co).

Greenfield, Lawrence A. 1996. <u>Child Victimizers: Violent Offenders and Their Victims</u>, Bureau of Justice Statistics, NCJ-153258, http://www.ncjrs.org./txtfiles/chldvrpt.txt.

Institute of Psychological and Educational Research (IPER). 1990. <u>Child Exploitation and</u> <u>Abuse: A Global Phenomenon</u>. A report of the first Asian Conference on child sexual exploitation and abuse. (Calcutta: Sree Saraswaty Press).

Snyder, H. N. & M. Sickmund. <u>Juvenile Offenders and Victims: A National Report</u> (Washington: Office of Juvenile Justice and Delinquency Prevention).

United Nations. 1995. <u>World Situation With Regard to International Traffic in Minors.</u> (Costa Rica: UN Latin American Institute for the Prevention of Crime and the Treatment of Offenders).

Section D. Impact of Sexual Exploitation on Children

Brannigan, Augustine & Erin Gibbs Van Brunschot. 1997. Youth prostitution and child sexual trauma, <u>International J of Law & Psychiatry</u> 20(3):337-354.

Cevallos, Diego. 1998. Rights-Mexico: Sterile at age 12, AIDS at 14, (February 10), [http://www.oneworld.org/ips2/feb98/mexico2.h tml].

Chan, C. S. 1987. Asian-American women: Psychological responses to sexual exploitation and cultural stereotypes, <u>Asian-American Psychological Association J</u> 12:11-15.

Hughes, Donna M. and Claire M. Roche (Editors). 1999. <u>Making the Harm Visible: Global</u> <u>Sexual Exploitation of Women and Girls--</u> <u>Speaking Out and Providing Services</u> (Kingston RI: The Coalition Against Trafficking in Women).

Joseph, Cathy. 1995. Scarlet wounding: Issues of child prostitution, <u>J of Psychohistory</u> 23(1):2-17.

Molnar, Beth E. et al. 1998. Suicidal behavior and sexual/physical abuse among street youth, <u>Child Abuse and Neglect</u> 22(3):213-222.

Molnar, Beth F, Stephen L. Buka & Ronald C. Kessler. 2001. Child sexual abuse and subsequent psychopathology: Results from the national comorbidity survey, <u>American Journal of</u> <u>Public Health</u> 91(5):753-760.

Mullen, P. E. et al. 1996. The long-term impact of the physical, emotional, and sexual abuse of children: A community study, <u>Child Abuse and</u> <u>Neglect</u> 20(1):7-20.

Pyett, P. M. & D. J. Warr. 1997. Vulnerability on the streets: Female sex workers and HIV risk, <u>AIDS Care</u> 9(5):539-547. Simons, R.L. & L.B. Whitbeck. 1991. Sexual abuse as a precursor to prostitution and victimization among adolescent and adult homeless women, Journal of Family Issues 12:361-379.

Section E. Impact of Sexual Exploitation on Adult Survivors

Finkelhor, David et al. 1990. Sexual abuse in a national survey of adult men and women: Prevalence, characteristics and risk factors, <u>Child</u> Abuse and Neglect 14:19-28.

Kilpatrick, Dean & Benjamin Saunders. 1997. The prevalence and consequences of child victimization, <u>Research Preview</u> (of the NIJ), April, pp. 1-4.

Widom, C.S. & S. Morris. 1997. Accuracy of adult recollections of child victimization: Part II, <u>Psychological Assessment</u> 9:34-46.

Wind, T.W. & L. Silvern. 1992. Type and extent of child abuse as predictors of adult functioning, Journal of Family Violence 7:261-281.

Section F. Forms of Child Sexual Exploitation

1. Child Pornography

Burgess, Ann W. & Marieanne L. Clark (Editors). 1984. <u>Child Pornography and Sex Rings</u> (New York: Free Press).

Editor. 1998. Fourteen nations join to bust huge internet child porn ring, <u>CNN Interactive</u> <u>http://www.cnn.com/WORLD/europe/9809/02/i</u> <u>nternet.porn.02/</u>.

Freeh, Louis J. 1997. <u>Child Pornography</u>, Public statement by the Director of the Federal Bureau of Investigation Before the Senate Appropriations Committee, Subcommittee on the Departments of Commerce, Justice, and State, the Judiciary, and Related Agencies Committee on the States Senate, Washington, D.C., April 8.

Gam, Phil. 1996. Project *Special Delivery*: Postal Inspectors Bust Child Porn Ring--and its Patrons, <u>U.S. Postal Inspection Service Bulletin</u> (October).

Goldstein, Michael J. 1973. <u>Pornography and</u> <u>Sexual Deviance</u> (Berkeley: University of California Press).

Healy, Margaret A. 1995. <u>Child Pornography:</u> <u>An International Perspective</u>. Prepared for the World Congress Against Commercial Sexual Exploitation of Children, Stockholm, Sweden <u>http://193.135.156.14/webpub/csechome</u>.

Leuchtag, Alice. 1995. The culture of pornography, <u>The Humanist</u> 55(3):4-7.

O'Brien, Shirley. 1992. <u>Child Pornography</u> (Dubuque: Kendall/Hunt Publishing Company).

O'Neil, Carl F. 1993. <u>Pornography</u> (New York: William Gladden Foundation).

Pierce, Robert L. 1984. Child pornography: A hidden dimension of child abuse, <u>Child Abuse</u> and Neglect 8(4):483-493.

Silbert, Mimi H. 1989. The effects on juveniles of being used for pornography and prostitution, in Zillmann, Dolf and Jennings Bryant (Eds.). <u>Pornography: Research Advances and Policy</u> <u>Considerations</u> (Hillsdale, NJ: Lawrence Erlbaum Associates, Inc.), pp. 215-234.

Silbert, Mimi H., Ayala M. Pines. 1984. Pornography and sexual abuse of women, <u>Sex Roles</u> 10(11-12):84, 857-868.

Svedin, Carl & Kristina Black. 1997. <u>Children</u> Who Don't Speak Out: About Children Being

Page 28 of 42

<u>Used in Child Pornography</u> (Bristol: Taylor & Francis).

Tate, Tim. 1991. <u>Child Pornography: An Inves-</u> tigation (North Pomfret: Trafalgar Square).

Wallace, B.D. & M.J. Philippus. 1990. <u>Children</u> of Pornography (Kettering: PPI Publishing).

2. Child Prostitution

Able-Peterson, Trudee. 1981. <u>Children of the</u> <u>Evening</u> (New York: Putnam & Sons).

Araji, S. 1997. <u>Sexually Aggressive Children:</u> <u>Coming to Understand Them</u> (Thousand Oaks CA: Sage).

Baizerman, M. M. et al. 1979. Adolescent prostitution, <u>Children Today</u> 8(September/October): 20-24.

Barrett, D. & W. Beckett. 1996. Child prostitution: Reaching out to children who seek sex to survive, <u>British J of Children's Nursing</u> 5(18):1120-1125.

Barry, Kathleen. 1995. <u>The Prostitution of Sexuality</u> (New York: New York University Press).

Brannigan & Gibbs van Brunschott. 1997. Youthful prostitution and child sexual trauma," International J of Law and Psychiatry 337

Brown, Marjorie E. 1979. Teenage prostitution, Adolescence 14(56):665-680.

Calcetas-Santos, Ofelia. 1997. <u>Report of the</u> <u>Special Rapporteur on the Sale of Children</u>, <u>Child Prostitution and Child Pornography on the</u> <u>Issue of Commerical Sexual Exploitation of</u> <u>Children in the United States of America</u> (New York: United Nations), Document #E/CN.4/1997/95/Add.2.

Dean, Ruth. 1997. <u>Teen Prostitution</u> (San Diego: Lucent Books).

Farley, Melissa. 2000. Getting real facts about prostitution (http://merge.simplenet.com/farley.html

Farley, Melissa. 2000. Prostituion: Factsheet on human rights violations (http://www.prostitutionresearch.com/factsheet.h tml)

Fassett, Byron & Bill Walsh. 1994. Juvenile prostitution: An overlooked form of child sexual abuse, <u>The APSAC Advisor</u> 7(1):9.

Giobbe, Evelina. 1990. Confronting liberal lies about prostitution, in Leidholdt, Dorchen & Janice G. Raymond (Editors), <u>The Sexual Liberals and the Attack on Feminism</u> (New York: Pergamon Press), pp. 67-82.

Giobbe, Evelina. 1992. Juvenile prostitution: Profile of recruitment, in Burgess, Ann W. (Editor). <u>Child Trauma: Issues and Research</u>, Volume 1 (New York: Garland Publishing Inc).

Greene, Jody M., Susan T. Ennett, and Christopher Ringwalt. 1999. Prevalence and correlates of survival sex among runaway and homeless youth, <u>American Journal of Public Health</u> 89(9):1406-1409.

Haq, Farquan. 1996. U.S. children: Street kids turn to sex to survive, <u>Inter-Press Service</u> (March 27).

Hofstede, Al. 1999. <u>The Hofstede Committee</u> <u>Report: Juvenile Prostitution in Minnesota</u> (Minneapolis).

Jaffe, Maureen & Sonia Rosen (Editors). 1996. Forced Labor: The Prostitution of Children (Upland: Diane Publishing Company).

Kane, June. 1998. <u>Sold For Sex</u> (Brookfield: Ashgate Publishing Co.).

Klain, Eva J. 1999. <u>Prostitution of Children and</u> <u>Child-Sex Tourism: An Analysis of Domestic</u> <u>and International Responses</u> (Arlington VA: National Center for Missing and Exploited Children). Mitchell, S. and L. Smith. 1984. <u>Juveniles in</u> <u>Prostitution: Fact vs. Fiction</u> (San Jose CA: R & E Publishers).

National Center for Missing and Exploited Children (NCMEC). 1998. <u>Child Prostitution</u>. (http:// <u>www.missingchildren.com</u>).

National Center for Missing and Exploited Children. <u>Female Juvenile Prostitution: Problem and</u> <u>Response</u>, (Arlington VA: NCMEC), 88 pp.

National Center for the Prosecution of Child Abuse. 1994. <u>Child Abuse Crimes: Child Prosti-</u> <u>tution</u> (Alexandria VA: American Prosecutors Research Institute).

O'Neill, Amy. 1997. Prostitute women now, in Scambler, Graham & Annette Scambler (Editors), <u>Rethinking Prostitution: Purchasing Sex in</u> <u>the 1990s</u> (London: Routledge).

Parker, Joe. 1998. How prostitution works (<u>http://www.prostituionresearch.com/parker-how.html</u>

Silbert, Mimi H. & Ayala M. Pines. 1982. Entrance into prostitution, <u>Youth and Society</u> 13(4):471-500.

Shaw, Ian & Ian Butler. 1998. Understanding young people and prostitution: A foundation for practice? <u>British J of Social Work</u> 28(2):177-196.

Stadiem, William. 1999. Lap of luxury: Sex clubs in Southern California, <u>Los Angeles</u> <u>Magazine</u> 44(6):90.

Terry, Sara. 1991. Sins of the innocent, <u>Rolling</u> <u>Stones</u> 616:67-70.

Weisberg, D. Kelly. 1985. <u>Children of the Night:</u> <u>A Study of Adolescent Prostitution</u> (New York: Free Press).

2a) <u>Special Issues Related to Male</u> <u>Prostitution</u>

Adams, Mark. 1999. <u>Hustlers, Escorts and Porn</u> <u>Stars: An Insiders Guide to Male Prostitution in</u> <u>America</u> (Las Vegas: Insiders Guide).

Allen, Donald M. 1980. Young male prostitutes: A psychosocial study, <u>Archives of Sexual Behavior</u> 9:399-426.

Bauserman, Robert & Bruce Rind. 1997. Psychological correlates of male child and adolescent sexual experiences with adults: A review of the non-clinical literature, <u>Archives of Sexual</u> <u>Behavior</u> 26(2):105-141.

Cates, Jim A. 1989. Adolescent male prostitution by choice, <u>Child and Adolescent Social</u> <u>Work J</u> 6(2):151-156.

Cates, J.A. and J. Markley. 1992. Demographic, clinical and personality variables associated with male prostitution by choice, <u>Adolescence</u> 27(107):695-706.

Deisher, R., G. Robinson, & D. Boyer. 1982. The adolescent female and male prostitute, <u>Pe-diatric Ann.</u> 11(10):819-825.

Earls, Christopher M. & Helene David. 1989. A psychosocial study of male prostitution, <u>Ar-</u> <u>chives of Sexual Behavior</u> 18(5):401-419.

Echols, Mike. 1996. <u>Tony's Boys: The Largest</u> <u>Case of Child Prostitution in U.S. History</u> (Amherst: Prometheus Books).

Indiana, Gary. 1994. <u>Rent Boy</u> (Baltimore MD: Serpent's Tail).

Itel, Joseph. 1998. <u>A Consumer's Guide to Male</u> <u>Hustlers</u> (Binghamton NY: Haworth Press).

Luckenbill, David F. 1986. Deviant career mobility: The case of male prostitutes, <u>Social Problems</u> 33(4):284-297.

McCamish, Malcolm. 1999. The friends thou hast: Support systems for male commercial sex

workers in Pattaya, Thailand, <u>J of Gay and Les-</u> bian Social Services 9(2-3):161-191.

McNamara, R.P. 1994. The Times Square Hustler: Male Prostitution in New York City (Westport CT: Praeger).

Pleak, R. R. and H. Meyer-Bahlburg. 1990. Sexual behavior and AIDS knowledge of young male prostitutes in Manhattan. <u>J of Sex Research</u> 27(4):557-587.

Remafedi, G. 1990. Fundamental issues in the care of homosexual youth, <u>Medical Clinics of</u> North America 74(5):1169-1179.

Rettenmund, Matthew. 1997. <u>Boy Culture</u> (New York: St. Martin's Press).

Savin-Williams, R.C. 1994. Verbal and physical abuse and stressors in the lives of lesbian, gay male and bisexual youths: Association with school problems, running away, substance abuse, prostitution and suicide, J of Consulting and Clinical Psychology 62(2):261-269.

Simon, P.M. 1994. HIV and young male street prostitutes: A brief report, <u>J of Adolescence</u>.

Snell, C. L. 1995. <u>Young Men in the Street:</u> <u>Help-Seeking Behavior of Young Male Prosti-</u> <u>tutes</u>. (Westport CT: Praeger).

Waldorf, D.J. 1990. Needle sharing among male prostitutes: Preliminary findings of the Prospero Project, <u>J of Drug Issues</u> 20(2):390-334.

2b) <u>Special Issues Related to Female</u> <u>Prostitution</u>

Bryan, James H. 1965. Apprenticeships in prostitution, <u>Social Problems</u> 12(3):287-297.

Farley, Melissa and Vanessa Kelly. 2000. Prostitution: A critical review of the medical and social science literature, <u>Women and Criminal Justice</u> 11(4). Gibsonainyette, L. et al. 1988. Adolescent female prostitutes, <u>Archives of Sexual Behavior</u> 17.

Goodson, Patricia, Alexandra Evans & Elizabeth Edmundson. 1997. Female adolescents and onset of sexual intercourse: A theory-based review of research from 1984 to 1994, <u>J of Adolescent</u> <u>Health</u> 21(3):147-156.

Gurisik, U. E. 1997. Female prostitution, in Welldon, Estela V. and Cleo Van Velsen, eds. <u>A</u> <u>Practical Guide to Forensic Psychotherapy</u> (London: Jessica Kingsley Publishers), pp. 182-187.

Harris, Sara. 1960. <u>They Sell Sex</u>. (Greenwich: Fawcett).

Harway, Michele & Marsha B. Liss. 1999. Dating violence and teenage prostitution: Adolescent girls' interactions with the legal system, in Johnson, N. G. et al. (Editors), <u>Adolescent Girls:</u> <u>Strengths and Successes</u> (Washington DC: American Psychological Association).

Hood-Brown, Marcia. 1998. Trading for a place: Poor women and prostitution, <u>Journal of Poverty</u> 2(3):13-33.

Inciardi, James. 1984. Little girls and sex: A glimpse at the world of the 'baby pro', <u>Deviant</u> <u>Behavior</u> 5:71-78.

James, Jennifer, Nanette Davis & Peter Vitaliano. 1982. Female sexual deviance: A theoretical and empirical analysis, <u>Deviant Behavior</u> 3(2):175-195.

Jesson, Jill. 1993. Understanding adolescent female prostitution: A literature review, <u>British J</u> of Social Work 23(5):517-530.

McClanahan, S.F. et al. 1999. Pathways into prostitution among female jail detainees and their implications for mental health services, <u>Psychiatric Services</u> 50(12):1606-13.

Mueck, M. 1992. Mother sold food, daughter sells her body: The cultural continuity of prostitution, <u>Social Science Medicine</u> 35(7).

2c) <u>Special Issues Related to Prostitu-</u> tion and Gay, Lesbian, Bisexual and Transgender Youth

Arey Doug. 1995. Gay males and sexual abuse, in Lisa Aronson Fontes (Ed.) <u>Sexual Abuse in</u> <u>Nine North American Cultures</u> (Thousand Oaks, CA: Sage).

Boyer, Debra. 1989. Male prostitution and homosexual identity, Journal of Homosexuality 17(1-2):151-184.

Green, J. and L. Brinkin. 1994. <u>Investigation</u> <u>Into Discrimination Against Transgendered Peo-</u> <u>ple</u> (San Francisco: City and Country of San Francisco Human Rights Commission).

Kruks, Gabe. 1991. Gay and lesbian homeless/street youth: Special issues and concerns, <u>J</u> of Adolescent Health 12:515-518.

Rotheram-Borus, Mary Jane, et al. 1992. Lifetime sexual behaviors among predominately minority male runaways and gay/bisexual adolescents in New York City, <u>AIDS Education and</u> <u>Prevention</u> (Fall):34-42.

Treichel, Christa J. and Carol J. Zierman. 1999. Snapshots: Gay, Lesbian, Bisexual and Transgender Youth and Prostitution. (November).

2d) <u>Child Prostitution in Historical</u> <u>Perspective</u>

Alexander, Ruth M. 1995. <u>The "Girl Problem":</u> <u>Female Sexual Delinquency in New York: 1900-</u> <u>1930</u> (Ithaca: Cornell University Press). Frederique Delacoste & Alexander, Prisilla (Editors). 1987. <u>Sex Work: Writings By Women in</u> <u>the Sex Industry</u> (Pittsburgh: Cleis Press).

Barry, Kathleen. 1995. <u>The Prostitution of Sexuality</u> (New York: New York University Press).

Ashby, LeRoy. 1984. <u>Saving the Waifs: Re-</u> formers and Dependent Children, 1890-1917 (Philadelphia: Temple University Press).

Beard, Mary & John Henderson. 1997. With this body I thee worship: Sacred prostitution in antiquity, <u>Gender & History</u> 9(3):480-503.

Jeffreys, Sheila. 1997. <u>The Idea of Prostitution</u> (Melbourne: Spinifex Press).

Karras, Ruth M. 1998. <u>Common Women: Prosti-</u> <u>tution and Sexuality in Medieval England</u> (New York: Oxford University Press).

Katyal, Neal Kumar. 1993. Men who own women: A thirteenth amendment critique of forced prostitution, <u>The Yale Law Review</u> 103(3):791-826.

Odem, Mary E. 1997. <u>Delinquent Daughters:</u> <u>Protecting and Policing Adolescent Female</u> <u>Sexuality in the United States, 1885-1920</u> (Chapel Hill: University of North Carolina Press).

Pearson, Michael. 1972. <u>The Age of Consent:</u> <u>Victorian Prostitution and Its Enemies</u> (London: David and Charles).

Plachy, Sylvia & James Ridgeway. 1996. <u>Red</u> <u>Light: Inside the Sex Industry</u> (New York: Powerhouse).

Roberts, Nickie. 1993. <u>Whores in History</u> (New York: Harper Collins).

Rossiaud, Jacques (translated by Lydia Cochrane). 1988. <u>Medieval Prostitution</u>. (New York: Blackwell).

Sanders, Wiley B. 1970. <u>Juvenile Offenders for</u> <u>a Thousand Years</u>. (Chapel Hill: University of North Carolina Press)

Seagraves, Ann. 1994. <u>Soiled Doves: Prostitutes</u> <u>in the Early West</u> (Hayden ID: Wesanne Publications).

Shoemaker, Donald J. 1977. The teeniest trollops: 'baby pros,' 'chickens' and child prostitutes, in Clifton D. Bryant (Editor) <u>Sexual Deviancy</u> <u>Social Context</u> (New York: New Viewpoints), pp. 241-53.

Walkowitz, Judith R. 1982. <u>Prostitution and</u> <u>Victorian Society</u> (Cambridge: Cambridge University Press).

2e) <u>Child Prostitution in International</u> <u>Perspective</u>

Ali, Salma. 1996. <u>Movement Against Flesh Tra-</u> <u>de</u> (Bangladesh: National Women Lawyers Association).

Altink, Sietske. 1995. <u>Stolen Lives: Trading</u> <u>Women Into Sex and Slavery</u> (London: Scarlet Press).

Altman, Dennis. 2001. <u>Global Sex</u> (Chicago: University of Chicago Press).

Azaola, Elena, Gerardo González, et al. 1993. <u>El</u> maltrato y el abuso sexual a menores: Una aproximación a estos fenómenos en México, (México DF: Universidad Autónoma Metropolitana - Unicef - Covac).

Azaola, Elena, Karla Nieto y Laura Valverde. 1996. <u>La prostitución en la zona de La Merced,</u> <u>Fundación Ford - Centro de Investigaciones y</u> <u>Estudios Superiores en Antropología Social</u> (inédito).

Azaola, Elena. 1998. Los niños-paisaje, en: Derechos Humanos y Ciudadanía, Suplemento 22, La Jornada, 16-VII-1998:2,8. Azaola, Elena. 1998. <u>Prostitución infantil, en: IV</u> <u>Informe sobre los derechos y la situación de la</u> <u>infancia en México 1994-97</u>, (México DF: Colectivo Mexicano de Apoyo a la Niñez).

Azaola, Elena. 2001. <u>Stolen Childhood. Girl and</u> <u>Boy Victims of Sexual Exploitation in Mexico</u>. (México DF: UNICEF).

Bales, Kevin. 1999. <u>Disposable People: New</u> <u>Slavery in the Global Economy</u> (Berkeley: University of California Press).

Beyer. 1996. Child prostitution in Latin America, in Jaffe, Maureen & Sonia Rosen (Editors), Forced Labor: The Prostitution of Children (Washington: U.S. Department of Labor).

Cadet, Jean-Robert. 1998. <u>Restavec: From Hai-</u> <u>tian Slave Child to Middle-Class American</u> (Austin: University of Texas).

Calcetas-Santos, Ofelia. 1998. <u>Report of the</u> <u>Special Rapporteur on the Sale of Children,</u> <u>Child Prostitution and Child Pornography on the</u> <u>Issue of Commerical Sexual Exploitation of</u> <u>Children to Mexico</u> (New York: United Nations), Document #E/CN.4/1998/101/Add.2.

Calcetas-Santos, Ofelia. 1999. <u>Report of the</u> <u>Special Rapporteur on the sale of children, child</u> <u>prostitution and child pornogrphy</u>, United Nations, E/CN.4/1999/71.

Comisión Europea. 1996. <u>Comisión sobre la lucha contra el turismo sexual que afecta a niños</u> (Bruselas, Comisión Europea, 27-XI-96, CO-M(96) 547 final).

Chejter, Silvia. 2000. <u>Estudio sobre Explotación</u> <u>Sexual Comercial Infantil en Argentina</u> (Buenos Aires: UNICEF).

Dank, Barry M. Sex <u>Work and Sex Workers</u> (New Brunswick NJ: Transaction Publishers).

D'Asaro, Andrea & Foley, Dylan. 1997. Girls and the business of sex (Vietnam), <u>On the Issues</u> 6(3):26 ff. End Child Prostitution, Child Pornography, and the Trafficking of Children for Sexual Exploitation (ECPAT). 1996. <u>Europe and North America</u> <u>Regional Profile</u> (issued by the World Congress Against Commercial Sexual Exploitation of Children, held in Stockholm, Sweden, August 1996, p.70.).

Espacios de Desarrollo Integral (EDIAC), Comisión de Derechos Humanos del Distrito Federal (CDHDF) y UNICEF. 1996. <u>Al otro lado de</u> <u>la calle. Prostitución de menores en La Merced.</u> (México DF: CDHDF - UNICEF).

Farley, Melissa. 1998. Prostitution in five countries: Violence and post-traumatic stress disorder, Feminism & Psychology 8(4):405-426.

Franco Guzmán, Ricardo. 1973. <u>La prostitución</u> (México DF: Diana).

Gomezjara, Francisco y Estanislao Barrera. 1992. <u>Sociología de la prostitución</u>. (México D.F: Fontamara).

González, Gerardo, E. Azaola, *et.al.* 1993. El maltrato y el abuso sexual a menores: una aproximación a estos fenómenos en México (UAM – UNICEF – COVAC).

Görian Svedin, Carl & Kristina Back. 1996. Children who don't speak out. About children being used in child pornography. (Estocolmo: Räda Barnen).

Instituto de la Mujer. 1988. <u>La prostitución de las mujeres</u> (Madrid: Instituto de la Mujer).

Instituto Interamericano del Niño. 2000. <u>Violencia y Explotación Sexual contra Niños y Niñas</u> <u>en América Latina y el Caribe</u> (Montevideo: IIN-OEA).

International Bureau for Children's Rights (IBCR). 1999. <u>International Dimensions of the Sexual Exploitation of Children. Global Report</u>. (Montreal: IBCR).

Joshi, Mahdu D. (Editor). 1997. <u>Women and</u> <u>Children in Prostitution: Human Rights Perspec-</u> <u>tives</u> [India] (Columbia: South Asia Books). Kempadoo, Kamala and Jo Doezema (Editors). 1998. <u>Global Sex Workers: Rights, Resistance,</u> and Redefinition (London: Routledge).

Lagarde, Marcela. 1993. <u>Los cautiverios de las</u> <u>mujeres: madresposas, monjas, putas, presas y</u> <u>locas (México DF: Universidad Nacional Autó-</u> noma de México).

Lamas, Marta. 1993. El fulgor de la noche: Algunos aspectos de la prostitución callejera en la Ciudad de México, <u>Debate Feminista</u> 4(8).

Leñero, Luis. 1998. <u>Los niños de la y en la calle.</u> <u>Problemática y estrategias para abordarla (México DF: Academia Mexicana de Derechos</u> Humanos - Unión Europea).

Lim, Lin Lean. 1998. <u>The Sex Sector: The Economic and Social Bases of Prostitution in Southeast Asia</u> (Geneva: International Labour Organization).

MacInnes, R. 1994. <u>Children in the Game</u> (Calgary: Street Teams).

Mathews, F. 1989. <u>Familiar Strangers: A Study</u> of Adolescent Prostitution (Toronto: Central Toronto Youth Services).

Mayomo, Alakok. 1998. Rights-Tanzania: Children drawn into sex trade, (April 27),

Muntarbhorn, Vitit. 1996. International perspectives and child prostitution in Asia, in Jaffe, Maureen & Sonia Rosen (Editors). Forced Labor: The Prostitution of Children (Washington: U.S. Department of Labor).

Narvesen, O. 1989. <u>The Sexual Exploitation of</u> <u>Children in Developing Countries</u> (Oslo: Redd Barna).

Negrete, Norma. 2000. <u>México. En: Instituto</u> Interamericano del Niño. Violencia y Explotación Sexual contra Niños y Niñas en América latina y el Caribe, (Montevideo: IIN): 225-262. O'Connell, Davidson. 1998. <u>Prostitution, Power</u> and Freedom (Ann Arbor: University of Michigan Press).

Odzer, Cleo. 1994. <u>Patpong Sisters: An American Women's View of the Bangkok Sex World</u> (New York: Arcade Publishing).

Organización Mundial de la Salud (WHO). 1996. <u>Comercial Sexual Exploitation of Chil-</u> <u>dren: The Health and Psychological Dimensions</u> (Geneva: WHO).

Plant, Martin A. <u>AIDS, Drugs, and Prostitution</u> (London: Routledge Press).

Randon, R. K. & K.N. Sudarshan. 1997. <u>Child</u> <u>Prostitution</u> [India] (Columbia: South Asia Books).

Save the Children. 1996. Kids for Hire. (London: Save the Children).

Save the Children. 2000. <u>Sacred Lives: Cana-</u> <u>dian Aboriginal Children and Youth Speak Out</u> <u>About Sexual Exploitation</u> (Toronto: Save the Children).

Seneviratne, Maureen (Editor). 1994. <u>An Evil</u> <u>Under the Sun : The Sexual Exploitation of</u> <u>Children in Sri Lanka</u>. (Mt. Lavinia, Sri Lanka: Protecting Environment and Children Everywhere).

Sereny, Gitta. 1985. <u>The Invisible Children:</u> <u>Child Prostitution in America</u>, West Germany and Great Britain (New York: Knopf).

Shamim, Ishrat. 1993. <u>Homeless and Powerless:</u> <u>Child Victims of Sexual Exploitation</u>. (Dhaka, Bangladesh: Bangladesh Sociology Association, University of Dhaka).

Treguear L. Tatiana. 1994. <u>Niñas Prostituídas.</u> <u>Caso Costa Rica</u> (San José CR: Fundación Procal).

U.S. Department of Labor . 1995. "The (Child) Sex Industry," <u>By the Sweat & Toil of Children</u> <u>Vol. II</u>, (Washongton: U.S.D.o.L., Bureau of International Affairs), pp. 133-144. U.S. Department of Labor . 1996. <u>Forced Labor:</u> <u>The Prostitution of Childrem: Symposium Pro-</u> <u>ceedings</u> (Washongton: U.S.D.o.L., Bureau of International Affairs).

U.S. Department of Labor . 1998. "Commercial (Child) Sex Workers," <u>By the Sweat & Toil of</u> <u>Children Vol. V</u>, (Washongton: U.S.D.o.L., Bureau of International Affairs), pp. 31-33.

United Nations. 1994. <u>Promotion and Protection</u> of the Rights of Children: Sale of Children, child prostitution and child pornography: Note by the <u>Secretary-General</u>, (New York: UN General Assembly), Doc. A/49/478, October 5, p. 29.

Uribe, Patricia et al. 1995. Prostitución y SIDA en la Ciudad de México. <u>Salud Pública</u> 37(6):592-601.

3. Trafficking in Children for Sex

Abbott, Karen. 2000. Denver man accused of importing boys for sex, <u>Denver Sun-Times</u> February 26.

Barr, C. W., Clayton, M., Epstein, J., Ingwerson, M., Matloff, J. 1996. <u>Child Sex Trade: Battling a</u> <u>Scourge.</u> (Boston MA: Christian Science Publishing).

Booth, William. 1999. 13 charged in gang importing prostitutes, <u>Washington Post</u> August 21, pp. A3.

Booth, William. 1999. Prostitution rings used immigrants: Gangs ran brothels using Asian women, <u>Washington Post</u> August 21.

Budapest Group. 1999. <u>The Relationship Be-</u> tween Organized Crime and Trafficking in <u>Aliens</u> (Austria: International Centre for Migration Policy Development).

Caldwell, Gillian, Steven Galster, and Nadia Steinsor. 1997. <u>Crime & Servitude: An Expose</u> of the Traffic in Women for Prostitution from the Newly Independent States. (Moscow: Global Survival Network).

Campagna, Daniel S. 1988. <u>The Sexual Traffick-</u> ing in Children: An Investigation of the Child <u>Sex Trade</u> (Dover MA: Auburn House).

Chidley, Joe et al. 1996. Fighting the child sex trade: New focus on an ancient evil, <u>World Press</u> <u>Review</u> 43(11):6-7.

Freed, Dale and Cal Miller. 1999. Foreign women forced to be hookers: 150 charged after probe of strip clubs, <u>Toronto Star</u> July 21.

French, Howard. 1992. For the world's brothels, Caribbean daughters, <u>New York Times</u> April 20: A8.

Godfrey, Tom. 1999. Canada's sex trade pipeline? <u>Toronto Sun</u>, March 29: 1.

Godfrey, Tom. 1999. Korean sex slaves nabbed at U.S. border, <u>Toronto Sun</u>, March 19: 46.

Goldberg, Carey. 1995. Sex slavery, Thailand to New York; Thousands of indentured Asian prostitutes may be in U.S., <u>New York Times</u> September 11: B1.

Gutierrez, Estrella. 1998. Rights: Child traffic in Venezuela tip of the iceberg (January 11), [URL:

http://www.oneworld.org/ips2/jan98/venezuela2. html].

Ha, Julie. 1999. Mexican girl tells of being recruited for prostitution, <u>Los Angeles Times</u> July 2.

Hawaleshka, Danylo. 1997. Sex, lies and money: An 11-month investigation uncovers an Asian sex-slave ring (in suburban Toronto), <u>Maclean's</u> 110(38):24-25.

Kilias, M. et al. 1993. "Cross-Border Crime," European J of Criminal Policy and Research, 1:7-134. Leuchtag, Alice. 1995. Merchants of flesh, <u>The</u> <u>Humanist</u> 55(2):11-17.

Lamberti, Rob. 1998. Cops raid den of Thai sex slaves, <u>Toronto Sun</u>, May 10: 7.

Lamberti, Rob. 1998. Sex slaves: Fodder for flesh factories, <u>Toronto Sun</u>, May 10:46.

Lederer, Laura. 2001. <u>Human Rights Report on</u> <u>Trafficking of Women and Children</u> (Baltimore MD: Johns Hopkins University, The Paul H. Nitze School of Advanced International Studies).

Loney, Jim. 1998. U.S. indicts 16 in Mexican prostitution ring, <u>Reuters</u>, April 23.

McDonald, R. Robin. 1999. Atlanta house linked to prostitution ring; FBI: Asian girls used as sex slaves, <u>Atlanta Journal and Constitution</u>, March 26:F6.

McCormick, Erin and Jim Zamora. 2000. Slave trade still alive in U.S., <u>San Francisco Examiner</u>, February 13.

Miller, Ron. 1995. Fighting for my daughter, Knight-Ridder/Tribune News Service, January 4.

Navarro, Mireya. 1998. Group forced illegal aliens into prostitution, U.S. says, <u>New York Times</u>, April 24:A10.

O'Neil-Richard, Amy. 1999. International Trafficking in Women (and children) to the United States: A Contemporary Manifestation of Slavery and Organized Crime (Washington: Center for the Study of Intelligence of the U.S. Department of State).

Pope, Victoria. 1998. Trafficking in women, U.S. News and World Report 122(13):38ff.

Reinhardt, Emma & Charles Jacobs. 2000. A secret slave trade survives in the United States, Boston Globe (November 27).

Rosin, Hanna. 2000. Thai boy caught up in fight against sex trafficking, <u>Washington Post</u> May 15: A2.

Skrobanek, Siripon et al. 1997. <u>The Traffic in</u> <u>Women: Human Realities of the International</u> <u>Sex Trade</u> (New York: Zed Books).

Stefanova, Kristina. 1999. Six sentenced in U.S.-Mexico sex-slave ring, <u>Reuters</u>, April 16.

Venetis, Penny. 1997. International sexual slavery, <u>Women's Rights Law Reporter</u> 18(3):268-70.

United Nations. 1995. <u>World Situation with Regard to International Traffic in Minors</u>. (Costa Rica: UN Latin American Institute for the Prevention of Crime and the Treatment of Offenders).

Wijers, Marjan and Lap-Chew, Lin. 1997. <u>Trafficking in Women: Forced Labor and Slavery-Like Practice in Marriage, Domestic Labor and Prostitution</u> (Utrecht: STV).

Williams, Phil (Editor). 1999. <u>Illegal Immigra-</u> tion and Commercial Sex: The New Slave Trade (London: Frank Cass Publishers).

Wilson, Tamar D. 1997. Trafficking and prostitution: The growing exploitation of migrant women from Central and Eastern Europe, <u>International Migration Review</u> 31(2):490ff.

Yoon, Youngik. 1995. International sexual slavery, <u>Touro International Law Review</u> (Spring): 417-436.

Yoon, Y. 1997. <u>International Sexual Slavery</u>. (Washington: CG Issue Overviews).

Youth Advocate Program International (YAPI). 1998. <u>Children for Sale: Youth Involved in Pros-</u> <u>titution, Pornography and Sex Trafficking</u> (Washington DC: YAPI).

4. Child Sex Tourism

Boye, R. 1996. Child sex tourism a booming business in poor countries, <u>The Philippine Star</u> (September 29).

Editors. 1996. Hitting sex tourists, <u>Maclean's</u> 109(36):22.

Harris, Bruce. 1997. Casa Alianza warns that Central America is new sex tourism destination, (November 18), [URL: http://www.casaalianza.org/news.shoot/shoot14.htm.

Hodgson, Douglas. Sex tourism and child prostitution in Asia: Legal responses and strategies, <u>Melbourne University Law Review</u> 412.

Ireland, Kevin. 1993. <u>'Wish you Weren't Here:'</u> <u>The Sexual Exploitation of Children and the</u> <u>Connection with Tourism and International</u> <u>Travel</u> (London: Save the Children Fund [UK]).

Katz, Nancie L. 1997. Sex tours abroad find US market: Rights group say women are exploited-or still children, <u>The Boston Globe</u> (June 15): E2.

Klain, Eva J. 1999. Prostitution of Children and Child-Sex Tourism: An Analysis of Domestic and International Responses (Arlington VA: National Center for Missing and Exploited Children).

Li, Vicki F.. 1995. Child sex tourism to Thailand: The role of the United States as a consumer country, <u>Pacific Rim Law and Policy J</u> 505.

O'Grady, R. 1992. <u>The Child and the Tourist</u> (Bangkok: End Child Prostitution in Asian Tourism [EPCAT]).

Pettman, Jan Jindy. 1997. Body politics: International sex tourism, <u>Third World Quarterly</u> 18(1):93-108.

Seabrook, J. 1997. <u>North-South Relations: The</u> <u>Sex Industry</u>. (Third World Network Features). Staebler, Martin. 1996. Tourism and children in prostitution, paper presented at the World Congress Against the Sexual Exploitation of Children held at Stockholm, August 27-31).

Vachss, Andrew. 1995. <u>Batman: The Ultimate</u> <u>Evil</u> (New York: Warner Books).

Woodiwiss, M. 1993. "Crime's Global Reach," in F. Pearce and M. Woodiwiss (Eds.), <u>Global</u> <u>Crime Connections</u>. (Houndsmills, UK: Macmillan).

World Sex Guide. 1998. <u>The World Sex Guide:</u> <u>A Research Project About Prostitution World-</u> wide (http://worldsexguide.org)

5. The U.S. Military and Child Sexual Exploitation

Moon, Katharine. 1997. <u>Sex Among Allies:</u> <u>Military Prostitution in US-Korean Relations</u> (New York: Columbia University Press).

Sturdevant, Saundra Pollock and Brenda Stolfuz. 1992. <u>Let the Good Times Roll: Prostitution and the U.S. Military in Asia</u> (New York: The New Press).

PART IV. Use of the Internet in Sex Crimes Involving Children

Agora. 1998. <u>Pedophilia & Internet: Old Obsessions and New Crusades</u>. (Romc, October 27) Conference, <u>http://www.agora.stm.it/pedfilia-internet/doc-eng.htm</u>.

Akdeniz, Yaman (Editor). 1998. <u>United States</u> <u>Sections of Regulation of Child Pornography on</u> the Internet: Regulation of Child Pornography on the Internet: Cases and Materials related to Child Pornography on the Internet, http://www.cyber-rights.org/reports/child.htm.

ECPAT. 1998. <u>Child Pornography on the Internet-Background Papers for Child Pornography</u> <u>on the Internet.</u> A position Paper for ECPAT International Meeting Lyon, France, May 28-29, <u>http://www.crin.org/iasc/sekiss.htm</u>.

Editors. 1997. <u>Internet Sting 'Nets' 1,500 Child</u> <u>Pornography Suspects</u> (October 2), <u>http://www</u>.lawstreet.com/journal/art/art1002cpo rn.html.

Freeh, Louis J. 1998. <u>Child Pornography on the</u> <u>Internet and Sexual Exploitation of Children</u>. (Washington: Federal Bureau of Investigation), March 10.

Hughes, Donna. 1999. <u>Pimps and Predators on</u> the Internet: Globalizing the Sexual Exploitation of Women and Children (New York: ECPAT-US).

Mehta, Michael D. 1998. Sex on the net: Regulation and control of pornography in the new wired world, in Alexander, Cynthia and Leslie A. Pal (Editors), <u>Digital Democracy: Policy and</u> <u>Politics in the Wired World</u> (New York: Oxford University Press).

NCH Action for Children. 1998. <u>Children on the</u> <u>Internet: Opportunities and Hazard</u>. CRIN: http://www.crin.org.

Unesco. 1999. <u>Sexual Abuse of Children, Child</u> <u>Pornography and Paedophilia on the Internet:</u> <u>An International Challenge</u>. Expert Meeting, Paris 18-19 January,1999, http://unesco.org/web.world/child _screen/conf_index.htm.

U.S. Department of Justice. 1997. <u>Use of Computers in the Sexual Exploitation of Children</u> (Washington: Office of Justice Programs--OJJDP).

Wiley, Stephen R. 1997. <u>Crimes Against Chil-</u> <u>dren Facilitated by the Internet</u>, A statement by the Chief of the Violent Crime and Major Offenders Section of the Federal Bureau of Investigation, Washington, D.C. November 7.

PART V.

Laws, the Legal Environment and Legal Resources Related to the Sexual Exploitation of Children

Badger, Ann. 1995. Legislation against child sex tourism, [http://www.oneworld.org/ news/world/ tourists.html].

Edwards. 1997. The legal regulation of prostitution: A human rights issues, in Scambler, Graham & Annette Scambler (Editors), <u>Rethinking</u> <u>Prostitution: Purchasing Sex in the 1990s</u> (London: Routledge)

Elstein, Sharon G. & Noy Davis. 1997. <u>Sexual</u> <u>Relationships Between Adult Males and Young</u> <u>Teen Girls: Exploring the Legal and Social Re</u><u>sponses</u> (Washington DC: American Bar Association, Center on Children and the Law).

Farrior, Stephanie. 1997. The international law on trafficking in women and children for prostitution: making it live up to its potential, <u>Harvard</u> <u>Human Rights J</u> (Spring), pp. 213 ff..

Grasso, D. L. 1994. <u>Criminal Child Sexual</u> <u>Abuse and Exploitation Laws in Eight Mid-</u> <u>Western States: Recommendations for Legisla-</u> <u>tive Change</u>. (Washington: USDHHS, National Center on Child Abuse and Neglect).

International Bureau for Children's Rights (IBCR). 1997. <u>Extraterritorial Legislation in Re-</u> sponse to the International Dimension of Child <u>Sexual Exploitation.</u> (Montreal: IBCR).

Kendall, Virginia. 1998. The lost child: Congress's inability to protect our teenagers," <u>Northwestern University Law Review</u> 92(4):1307-1315.

Lederer, Laura. 2001. <u>Human Rights Report on</u> <u>Trafficking of Women and Children</u> (Baltimore MD: Johns Hopkins University, The Paul H. Nitze School of Advanced International Studies).

Loken, Gregory A. 1987. Child prostitution, in NCMEC <u>Child Pornography and Prostitution:</u> <u>Background and Legal Analysis</u> (Arlington VA: National Center for Missing and Exploited Children).

MacKinnon. 1993. Prostitution and civil rights, Michigan J of Gender and Law 13.

Office of Juvenile Justice and Delinquency Prevention (OJJDP). 1997. <u>Federal Resources on</u> <u>Missing and Exploited Children: A Directory for</u> <u>Law Enforcement and Other Public and Private</u> <u>Agencies.</u> (Washington: Federal Agency Task Force for Missing and Exploited Children).

PART VI. Cooperation in Prosecuting Adults Committing Sexual Crimes Against Children

1. <u>Cooperation Between U.S. Public</u> and Private Organizations

Connelly, Helen & Ron Laney. 2001. Missing and exploited children's training program, <u>OJJDP Fact Sheet</u> (March), #06.

Dimenstein, Gilberto. 1996. The role of the media, in Jaffe, Maureen & Sonia Rosen (Editors). Forced Labor: The Prostitution of Children (Washington: U.S. Department of Labor).

Hirsch, M. & J. Bolas. 1996. <u>Street Outreach</u> <u>Curriculum</u> (New York: Empire State Coalition of Youth and Family Services).

Loken, G. 1986. The federal battle against child sexual exploitation: Proposal for reform, <u>Har-vard Women's Law J</u> 9(Spring):105-134.

Monto, Martin A. 1998. Holding men accountable for prostitution: The unique approach to sexual exploitation education project (SEEP), <u>Violence Against Women</u> 4(4):505-517.

National Center for Missing and Exploited Children. <u>A Report to the Nation (regarding State</u> <u>legislative accomplishments)</u>, (Arlington VA: NCMEC), 52 pp.

National Clearinghouse on Families and Youth (NCFY). 1997. <u>Ending the Sexual Exploitation</u> of Young People: A Community Approach (Silver Spring MD: NCFY).

Office of Juvenile Justice and Delinquency Prevention (OJJDP). 1997. <u>Special Joint Report:</u> <u>Federal Agency Task Force for Missing and Exploited Children.</u> (Washington: Office of Juvenile Justice and Delinquency Prevention).

Price, V. 1992. <u>Street Training Manual for Run-away and Homeless Youth Service Providers</u> (Boston: Bridge Over Troubled Waters, Inc.).

Saikaew, Ladda. 1996. A non-governmental organization perspective, in Jaffe, Maureen & Sonia Rosen (Editors). <u>Forced Labor: The Prostitution of Children</u> (Washington: U.S. Department of Labor).

Subcommittee to Investigate Juvenile Delinquency. 1978. <u>Protection of Children Against</u> <u>Sexual Exploitation</u>. (Washington DC: US GPO).

U.S. Conference of Mayors. 1992. <u>Reaching</u> <u>Youth in High-Risk Situations: Three Approaches</u> (Washington: U.S. Conference of Mayors).

Wahab, Stephanie. 1998. Let's talk about sex work: Feminisms, social work and the sex industry, <u>Dissertation Abstracts International</u> 58(12):4814-A.

Whitcomb, Debra. 1995. <u>Child Sexual Exploitation: Improving Investigations and Protecting</u> <u>Victims--A Blueprint for Action</u> (Washington: U.S. Department of Justice, Office of Justice Programs, Office for Victims of Crime). Whitcomb, Debra, Edward De Vos, and Barbara E. Smith. 1998. <u>Program to Increase Under-</u> <u>standing of Child Sexual Exploitation, Final Re-</u> <u>port</u> (Washington DC: Education Development Center, Inc. & American Bar Association, Center on Children and the Law).

Whitcomb, Debra & Julie Eastin. 1998. Joining Forces Against Child Sexual Exploitation: Models For a Multi-jurisdictional Team Approach (Washington: U.S. Department of Justice, office of Juvenile Justice and Delinquency Prevention).

2. <u>International Cooperation</u>

Asian Regional Initiative Against Trafficking (ARIAT). 2000. <u>Country Plan of the United</u> <u>States</u>.

Berkman, Eric T.c 1996. Responses to the international child sex tourism trade, <u>Boston College</u> <u>International & Comparative Law Review</u> 397.

Boonpala, Panudda. 1996. The role of the International Labor Organization, in Jaffe, Maureen & Sonia Rosen (Editors), <u>Forced Labor: The</u> <u>Prostitution of Children</u> (Washington: U.S. Department of Labor).

Council of Europe, European Committee on Crime Problems. 1992. Extraterritorial criminal jurisdiction, <u>Criminal Law Forum</u> 3(3):441ff.

Editor. 1996. Britain joins the campaign against child sex tourism, (July 25), [URL: http://www.oneworld.org/news/conorts/iul06.abi

http://www.oneworld.org/news/reports/jul96_chi ldren.html].

Focal Point on Sexual Exploitation of Children. 1999. <u>Sharing Strategic Planning and Action to</u> <u>Minimize Duplication and Competition Through</u> <u>the Enhancement of Coordination at All Levels</u> (Geneva).

Focal Point on Sexual Exploitation of Children. 1999. <u>Regional/Sub-Regional Collaborators</u> (Working Against the Sexual Exploitation of Children). (Geneva).

Page 40 of 42

Harris, Bruce. 1997. Casa Alianza announces new program for Nicaragua, (December 15), [URL: http://www.casaalianza.org/news.shoot/shoot21.htm].

Harris, Bruce. 1998. Casa Alianza co-operates with FBI and US Justice Department to trap American child abusers in Central America, (February 13), [URL: <u>http://www.casa-</u> <u>alianza.org/news.shoot/shoot27.htm].</u>

Harris, Bruce. 1998. The global march against child labour arrives to Central America, (April 15), [URL: <u>http://www.casa-</u> <u>alianza.org/news.shoot/shoot41.htm]</u>.

Harris, Bruce. 1998. Arrest of another American sex tourist in Costa Rica, (May 4), [URL: <u>http://www.casa-</u> <u>alianza.org/news.shoot/shoot46.htm].</u>

International Bureau for Children's Rights (IBCR). 1997. <u>Annual Report, 1996/97</u>. (Montreal: IBCR).

International Bureau for Children's Rights (IBCR). 1998. <u>Report of Public Hearings (on the</u> <u>International Dimensions of Child Sexual Ex-</u> <u>ploitation held) in Fortaleza, Brazil, 11-15 May,</u> <u>1998.</u> (Montreal: IBCR).

International Bureau for Children's Rights (IBCR). 1998. <u>Report of Public Hearing (on the</u> <u>International Dimensions of Child Sexual Ex-</u> <u>ploitation Held) in Bangkok, Thailand 17-21</u> <u>November, 1998.</u> (Montreal: IBCR).

International Center for Criminal Law Reform and Criminal Justice Policy. 1995. <u>Bibliography</u> on the Effect of Organized Crime on Women and Children (Canada).

International Criminal Police Organization (IN-TERPOL). 1997. <u>International Crime Statistics</u>. (St. Cloud, France: Interpol).

Healy, Margaret A. 1995. Prosecuting child sex tourists at home: Do laws in Sweden, Australia, and the US safeguard the rights of children as mandated by international law? <u>Fordham International Law J</u> 18:1852ff.

Hecht, Mark. 1997. Canada's Bill C-27, how does it compare to extraterritorial legislation in other countries? <u>Human Rights Tribune</u> 4(1):10ff.

Hodgson, Douglas. 1995. Combating the organized sexual exploitation of Asian children: Recent developments and prospects. <u>International J</u> of Law & the Family 9(1): 23-53.

Levan, Patricia D. 1994. Curtailing Thailand's child prostitution through an international conscience, <u>American University J of International</u> Law and Policy (Spring), 869ff.

McDonald, W. F. 1995. The globalization of criminology: The new frontier is the frontier, <u>Transnational Organized Crime</u>, 1:1-12.

Office of the National Commission on Women's Affairs and Mekong Region Law Center (MRLC). 1999. <u>Trafficking in Women and</u> <u>Children</u> (Bangkok: Amarin Printing and Publishing Public Co). IBSN: 974-272-027-4.

Todres, Jonathan. 1999. Prosecuting sex tour operators in U.S. courts in an effort to reduce the sexual exploitation of children globally, <u>Boston</u> <u>University Public Interest Law J</u> 9:1-23.

Williams, Phil. 1995. Transnational criminal organizations: Strategic alliances, <u>Washington</u> <u>Quarterly</u>, 18:57-72.